UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM S-8

REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

L3 TECHNOLOGIES, INC.

(Exact name of registrant as specified in its charter)

Delaware

13-3937436

(State or other jurisdiction of incorporation or organization)

(I.R.S. Employer Identification Number)

L3 Technologies, Inc. **600 Third Avenue** New York, New York 10016 (Address of Principal Executive Offices) (Zip Code)

> L3 Technologies Master Savings Plan (Full title of the plan)

Ann D. Davidson, Esq. Senior Vice President, General Counsel and Corporate Secretary L3 Technologies, Inc. 600 Third Avenue New York, New York 10016 (Name and address of agent for service)

(212) 697-1111

(Telephone number, including area code, of agent for service)

Copies of all notices, orders and communication to:

Mark Brod, Esq. Simpson Thacher & Bartlett LLP **425 Lexington Avenue** New York, New York 10017-3954 (212) 455-2000

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act:

Large accelerated filer \square

Accelerated filer o

Non-accelerated filer o (Do not check if a smaller reporting company) Smaller reporting company \Box

CALCULATION OF REGISTRATION FEE

Title of securities to be registered	Amount to be registered(1)	Proposed maximum offering price per share(2)	Proposed maximum aggregate offering price(2)	Amount of registration fee(2)
Common Stock, par value \$0.01 per share	3,000,000 shares	\$167.34	\$502,020,000	\$58,185

Securities Act of 1933, as amended (the "Securities Act"), the number of shares registered includes an indeterminable number of shares of Common Stock issuable under the L3 Technologies Master Savings Plan, as a result of stock splits, stock dividends, anti-dilution provisions or similar transactions.

Pursuant to Rule 457(c) and Rule 457(h) under the Securities Act, the proposed maximum offering price per share, the proposed maximum aggregate offering price and the amount of registration fee have been computed based on the average of the high and low prices per share of Common Stock reported on the New York Stock Exchange on February 16, 2017.

EXPLANATORY NOTE

This Registration Statement on Form S-8 (the "Registration Statement") is filed by L3 Technologies, Inc. (the "Company" or the "Registrant") to register an additional 3,000,000 shares of Common Stock under the L3 Technologies Master Savings Plan (the "Plan") for which previously filed registration statements on Form S-8 relating to the Plan are effective. Pursuant to General Instruction E to Form S-8, this Registration Statement incorporates by reference contents of the registration statements on Form S-8 filed by L-3 Communications Holdings, Inc. (as predecessor to the Registrant, the "Predecessor Registrant") with the Securities and Exchange Commission (the "Commission") on March 11, 2003 (File No. 333-103752), June 28, 2007 (File No. 333-144135), August 2, 2010 (File No. 333-168467) and August 16, 2012 (File No. 333-183353), as amended by Post-Effective Amendment No. 1 filed by the Registrant with the SEC on January 3, 2017, except to the extent supplemented, amended or superseded by the information set forth herein.

PART I

INFORMATION REQUIRED IN THE SECTION 10(a) PROSPECTUS

The information specified in Items 1 and 2 of Part I of this Form S-8 is omitted from this filing in accordance with the provisions of Rule 428 under the Securities Act and the introductory note to Part I of Form S-8. The documents containing the information specified in Part I have been or will be delivered to the participants in the Plan as required by Rule 428(b)(1) under the Securities Act. Such documents are not required to be, and are not, filed with the Commission either as part of this Registration Statement or as a prospectus or prospectus supplement pursuant to Rule 424 under the Securities Act.

PART II

INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

ITEM 3. INCORPORATION OF DOCUMENTS BY REFERENCE

The following documents filed with the Commission by the Company are hereby incorporated herein by reference:

- (a) Annual Report on Form 10-K of the Company for the year ended December 31, 2016;
- (b) Current Reports on Form 8-K of the Company filed on January 3, 2017 and January 25, 2017; and
- (c) The description of the Predecessor Registrant's common stock contained in the Predecessor Registrant's Registration Statement on Form 8-A12B filed on May 18, 1998, as amended by the description of the Predecessor Registrant's common stock included in the Predecessor Registrant's Registration Statement on Form S-3 filed on November 23, 2005 and as further amended by the descriptions of the amendments to the Predecessor Registrant's Amended and Restated Certificate of Incorporation as described on pages 22-25 of the Predecessor Registrant's definitive proxy statement for the Annual Meeting filed on March 18, 2013.

All documents filed by the Company pursuant to Sections 13(a), 13(c), 14 and 15(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act") (other than information furnished pursuant to Item 2.02 or 7.01 of any Current Report on Form 8-K, unless expressly stated otherwise therein) after the date hereof and prior to the filing of a post-effective amendment to this Registration Statement which indicates that all securities offered have been sold or which deregisters all securities then remaining unsold shall be deemed to be incorporated by reference herein and to be a part hereof from the date of filing such documents. Any statement contained in a document incorporated or deemed to be incorporated by reference herein shall be deemed to be modified or superseded for purposes hereof to the extent that a statement contained herein or in any other subsequently filed document which also is or is deemed to be incorporated by reference herein modifies or supersedes such statement. Any such statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute a part hereof.

ITEM 4. DESCRIPTION OF SECURITIES

Not applicable.

ITEM 5. INTERESTS OF NAMED EXPERTS AND COUNSEL

None.

ITEM 6. INDEMNIFICATION OF DIRECTORS AND OFFICERS

Section 145 of the Delaware General Corporation Law (the "DGCL") provides for, among other things:

- (i) permissive indemnification for expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by designated persons, including directors, officers, employees or agents of a corporation, in the event such persons are parties to litigation other than stockholder derivative actions if certain conditions are met:
- (ii) permissive indemnification for expenses (including attorneys' fees) actually and reasonably incurred by designated persons, including directors, officers, employees or agents of a corporation, in the event such persons are parties to stockholder derivative actions if certain conditions are met;
- (iii) mandatory indemnification for expenses (including attorneys' fees) actually and reasonably incurred by designated persons, including directors and officers of a corporation, in the event such persons are successful on the merits or otherwise in defense of litigation covered by (i) and (ii) above; and
- (iv) that the indemnification and advancement of expenses provided for by Section 145 is not deemed exclusive of any other rights which may be provided under any by-law, agreement, stockholder or disinterested director vote, or otherwise.

In addition to the indemnification provisions of the DGCL described above, the Registrant's Restated Certificate of Incorporation (the "Certificate of Incorporation") provides that the Registrant shall, to the fullest extent permitted by the DGCL, (i) indemnify its officers and directors and (ii) advance expenses incurred by such officers or directors in relation to any action, suit or proceeding.

The Registrant's Amended and Restated Bylaws (the "Bylaws") require, in certain instances, the advancement of expenses to an officer or director (without a determination as to his conduct) in advance of the final disposition of a proceeding if such person furnishes a written undertaking to repay any advances if it is ultimately determined that he is not entitled to indemnification.

The Bylaws purport to confer upon officers and directors contractual rights to indemnification and advancement of expenses as provided therein. The right to indemnification and advancement of expenses as provided therein shall (i) vest at the time that such claimant becomes a director or officer of the Registrant or at the time such claimant becomes a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust or other enterprise, including service with respect to employee benefit plans, at the request of the Registrant and (ii) continue as to the claimant even though he may have ceased to be a director or officer of the Registrant.

The Certificate of Incorporation limits the personal liability of directors to the Registrant or its stockholders for monetary damages for breach of the fiduciary duty as a director, other than liability as a director (i) for breach of duty of loyalty to the Registrant or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of the DGCL or (iv) for any transaction from which the director derived an improper personal benefit.

The Registrant maintains officers' and directors' insurance covering certain liabilities that may be incurred by officers and directors in the performance of their duties.

ITEM 7. EXEMPTION FROM REGISTRATION CLAIMED

Not applicable.

ITEM 8. EXHIBITS

For the list of exhibits, see the Exhibit Index hereto, which is incorporated into this item by reference.

ITEM 9. UNDERTAKINGS

- (a) The undersigned registrant hereby undertakes:
 - (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this Registration Statement:
 - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act;
 - (ii) To reflect in the prospectus any facts or events arising after the effective date of the Registration Statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the Registration Statement; and
 - (iii) To include any material information with respect to the plan of distribution not previously disclosed in the Registration Statement or any material change to such information in the Registration Statement;

Provided, *however*, that paragraphs (a)(1)(i) and (a)(1)(ii) do not apply if the information required to be included in a post-effective amendment by those paragraphs is contained in reports filed with or furnished to the Commission by the Registrant pursuant to section 13 or section 15(d) of the Exchange Act that are incorporated by reference in the Registration Statement.

- (2) That, for the purpose of determining any liability under the Securities Act, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (b) The undersigned registrant hereby undertakes that, for purposes of determining any liability under the Securities Act, each filing of the Registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Exchange Act) that is incorporated by reference in the Registration Statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.
- (c) Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the Registrant pursuant to the foregoing provisions, or otherwise, the Registrant has been advised that in the opinion of the Commission such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, as amended, the registrant certifies that it has reasonable grounds to believe that it meets all of the requirements for filing on Form S-8 and has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of New York, State of New York, on this 23rd day of February, 2017.

L3 TECHNOLOGIES, INC.

By: /s/ Ann D. Davidson

Name: Ann D. Davidson

Title: Senior Vice President, General Counsel and Corporate Secretary

POWER OF ATTORNEY

KNOW ALL PERSONS BY THESE PRESENTS, that each person whose signature appears below hereby constitutes and appoints each of Michael T. Strianese, Ralph G. D'Ambrosio, Ann D. Davidson, Esq. and Allen E. Danzig, Esq., as his or her true and lawful attorney-in-fact and agent, with full power of substitution and re-substitution, for him in his name, place and stead, in any and all capacity, in connection with this Registration Statement, including to sign and file in the name and on behalf of the undersigned as director or officer of the registrant any and all amendments or supplements (including any and all stickers and post-effective amendments) to this Registration Statement, with all exhibits thereto, and other documents in connection therewith with the Securities and Exchange Commission and any applicable securities exchange or securities self-regulatory body, granting unto said attorney-in-fact and agent, with full power and authority to do and perform each and every act and thing requisite or necessary to be done in and about the premises, as fully to all intents and purposes as he might or could do in person, hereby ratifying and confirming all that said attorney-in-fact and agents, or any substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, as amended, this Registration Statement has been signed below by the following persons in the capacities indicated on the 23rd day of February, 2017.

Signature	Title
/s/ Michael T. Strianese	Chairman and Chief Executive Officer
Michael T. Strianese	(Principal Executive Officer) and Director
/s/ Ralph G. D'Ambrosio	Senior Vice President and Chief Financial Officer
Ralph G. D'Ambrosio	(Principal Financial Officer)
/s/ Dan Azmon	Vice President, Controller and Principal Accounting Officer
Dan Azmon	(Principal Accounting Officer)
/s/ Claude R. Canizares	Director
Claude R. Canizares	75.
/s/ Thomas A. Corcoran	Director
Thomas A. Corcoran	
/s/ Ann E. Dunwoody	Director
Ann E. Dunwoody	
/s/ Lewis Kramer	Director
Lewis Kramer	
/s/ Robert B. Millard	Director
Robert B. Millard	
/s/ Lloyd W. Newton	Director
Lloyd W. Newton	
/s/ Vincent Pagano, Jr.	Director
Vincent Pagano, Jr.	
/s/ H. Hugh Shelton	Director
H. Hugh Shelton	
/s/ Arthur L. Simon	Director
Arthur L. Simon	

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, as amended, the trustees (or other persons who administer the Plan) have duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of New York, State of New York, on the 23rd day of February, 2017.

L3 TECHNOLOGIES MASTER SAVINGS PLAN

By: /s/ Ann D. Davidson

Name: Ann D. Davidson

Title: Senior Vice President, General Counsel and

Corporate Secretary

EXHIBIT INDEX

Exhibit Number	Description of Exhibit
4.1	Restated Certificate of Incorporation of L3 Technologies, Inc. (incorporated by reference to Exhibit 3.1 of the Company's Current Report on Form 8-K filed on January 3, 2017).
4.2	Amended and Restated Bylaws of L3 Technologies, Inc. (incorporated herein by reference to Exhibit 3.2 of the Company's Current Report on Form 8-K filed on January 3, 2017).
4.3	Form of Common Stock Certificate (incorporated herein by reference to Exhibit 4.1 of the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2016 (File No. 001-37975).
4.4	L3 Technologies Master Savings Plan*
5	Opinion of Simpson Thacher & Bartlett LLP, with respect to legality of securities being registered hereunder.*
23.1	Consent of PricewaterhouseCoopers LLP.*
23.2	Consent of Simpson Thacher & Bartlett LLP (included as part of Exhibit 5 filed herewith).*

*Filed herewith

${f E}$	xhibit 4.4
L3 TECHNOLOGIES MASTER SAVINGS PLAN	
(Effective January 1, 2017)	

TABLE OF CONTENTS

		<u>r age</u>
INTROD	DUCTION	1
ARTICL	E I - DEFINITIONS	2
1.	.1 Account	2
	.2 Affiliate	2
	.3 After-Tax Contributions	2
	.4 After-Tax Contribution Account	2 2
1.	.5 Beneficiary	2
1.	.6 Board of Directors	3
1.	.7 CATCH-UP CONTRIBUTIONS	3
1.	.8 CATCH-UP CONTRIBUTION ACCOUNT	3
1.	.9 Code	3
1.	.10 Committee	3
1.	.11 Company	3
1.	.12 Compensation	
1.	.13 Dividend Election	4
1.	.14 Eligible Employee	5
	.15 Employee	5
	.16 Employee Contributions	5
	.17 Employee Contribution Account	5
	.18 Employer	5
	.19 Employer Contribution Account	5 5
	.20 Employer Contributions	5
	.21 ERISA	6
	.22 ESOP	6
	.23 ESOP ACCOUNT	6
	.24 Former Participant	6
	.25 Highly Compensated Employee	6 7
	.26 Hour of Service	
	.27 Investment Fund	8
	.28 L3 Stock	8
	.29 L3 Stock Fund	8
	.30 Matching Contributions	8
	.31 Matching Contribution Account .32 Non-Covered Employer	8 9
	.33 Non-Covered Status	9
	34 Non-Highly Compensated Employee	9
	.35 Normal Retirement Date	9
	.36 PARTICIPANT	9
	.37 Period of Service	9
	.38 Period of Severance	9
	.39 Plan	9
	.40 Plan Year	9
	.41 Pre-Tax Contributions	9
	.42 Pre-Tax Contribution Account	10
	.43 Prior Plan	10
	.44 Recordkeeper	10
	.45 Rollover Contributions	10
	.46 ROLLOVER CONTRIBUTION ACCOUNT	10
1.	.47 Roth Elective Deferral Account	10
1.	.48 Service	10

L3 Technologies Master Savings Plan

TABLE OF CONTENTS (continued)

			<u>Page</u>
	1.49	SEVERANCE FROM SERVICE DATE	10
	1.50	Supplemental Contributions	11
	1.51	Supplemental Contribution Account	11
	1.52	Termination of Employment	11
	1.53	Total Disability	11
	1.54	Trust or Trust Fund	11
	1.55	Trust Agreement	11
	1.56	Trustee	11
	1.57	VALUATION DATE	11
			40
ARTI	JLE II -	ADMINISTRATION	12
	2.1	Сомміттее	12
	2.2	DISCRETIONARY POWER TO INTERPRET AND ADMINISTER THE PLAN	12
	2.3	GENERAL PROVISIONS	12
	2.4	Power to Execute Plan and Government Documents	13
	2.5	CLAIMS PROCEDURE	13
	2.6	Indemnification	13
ARTI	CLE III	- PARTICIPATION	14
	3.1	General Conditions of Eligibility	14
	3.2	Election to Participate	14
	3.3	Transfers from Non-Covered Status	14
	3.4	Transfer to Non-Covered Status	14
	3.5	Transfers Among Participating Employers	15
	3.6	ELIGIBILITY UPON RE-EMPLOYMENT	15
	3.7	Service Under Elapsed Time Method	15
	3.8	Qualified Military Service	16
	3.9	FMLA	16
ARTI	CLE IV	- CONTRIBUTIONS	17
		CONTRIBUTION	
	4.1	Pre-Tax Contributions	17
	4.2	After-Tax Contributions	18
	4.3	CATCH-UP CONTRIBUTIONS	19
	4.4	MATCHING CONTRIBUTIONS	19
	4.5	Other Employer Contributions	20
	4.6	ROLLOVER CONTRIBUTIONS	20
	4.7	Contributions Required by the terms of a Collective Bargaining Agreement.	21
	4.8	Suspension of Contributions Upon Transfer to Non-Covered Status	21
	4.9	Timing of Contributions to Trustee	21
	4.10	METHOD BY WHICH CONTRIBUTIONS ARE MADE TO THE TRUST	21
	4.11	Transfers from Prior Plan	21
	4.12	Qualified Non-Elective Contributions	22
ARTIO	CLE V -	LIMITATIONS ON CONTRIBUTIONS	23
	5.1	Suspension of Contributions Upon Reaching the Savings Maximum	23
	5.2	Return of Excess Deferrals	23
		ii	L3 Technologies Master Savings Plan

TABLE OF CONTENTS (continued)

			<u>Page</u>
	5.3	Section 401(k) Limit on Pre-Tax Contributions	24
	5.4	Section 401(m) Limit on Matching Contributions	25
	5.5	Annual Additions Limit	27
ART	ICLE VI	- PARTICIPANTS' ACCOUNTS	31
	6.1	Establishment of Accounts	31
	6.2	ACCOUNTS IN INVESTMENT FUNDS	31
	6.3	How Accounts are Valued	31
ART	ICLE VI	I - INVESTMENT OF CONTRIBUTIONS; TRANSFERS BETWEEN FUNDS	32
	7.1	Participant Directed Investments	32
	7.2	DISCONTINUED FUNDS	33
	7.3	Limitation or Suspension of Transaction and Limitation of Daily Securities Trading	33
ART	ICLE VI	II - VESTING	34
	8.1	Full Vesting in Employee Contribution Account	34
	8.2	Vesting in Employer Contribution Account	34
	8.3	Forfeitures	34
ART	ICLE IX	- WITHDRAWALS PRIOR TO TERMINATION OF SERVICE; LOANS	36
	9.1	Withdrawals	36
	9.2	WITHDRAWAL OF AFTER-TAX CONTRIBUTIONS	36
	9.3	WITHDRAWAL OF ROLLOVER CONTRIBUTION ACCOUNT	36
	9.4	WITHDRAWAL OF VESTED MATCHING CONTRIBUTION ACCOUNT	36
	9.5	WITHDRAWAL OF PRE-TAX CONTRIBUTIONS	36
	9.6	Hardship Withdrawals	36
	9.7	WITHDRAWAL OF CATCH-UP CONTRIBUTIONS	38
	9.8	WITHDRAWAL PRO-RATA FROM INVESTMENT FUNDS	38
	9.9	Timing of Withdrawal Payments	38
	9.10	Loans	38
A DO		DICTRIPIONO	44
ART	ICLE X	- DISTRIBUTIONS	41
	10.1		41
	10.1	PAYMENT UPON TERMINATION OF EMPLOYMENT	41
	10.2	CASH-OUT	41
	10.3	Application for Benefits	41
	10.4	General Rules	41
	10.5 10.6	Consent for Early Distributions Direct Rollover	42 42
		DISTRIBUTIONS IN CASH OR STOCK	
	10.7	Qualified Joint and Survivor Annuity	43 43
	10.8 10.9	QUALIFIED PRETEIREMENT SURVIVOR ANNUITY QUALIFIED PRETEIREMENT SURVIVOR ANNUITY	43
	10.9	QUALIFIED FRENETIKEMENT SURVIVUK ANNUITY	44

iii

TABLE OF CONTENTS (continued)

		<u>Page</u>
ARTICLE XI	- SPECIAL TOP-HEAVY PROVISIONS	46
11.1	Top-Heavy Rules	46
11.2	Definitions	46
11.3	MINIMUM CONTRIBUTION	48
11.4	Top-Heavy Vesting Schedule	48
ARTICLE XI	I - FUNDING OF THE SAVINGS PLAN; TRUST FUND	50
		_
12.1	Trust Agreement	50
	Income on Funds	50
	EXCLUSIVE BENEFIT OF TRUST FUND	50
	MISTAKE OF FACT	50
12.5	CONTRIBUTIONS DISALLOWED BY CODE	50
ARTICLE XI	II - AMENDMENT AND TERMINATION	51
13.1	PLAN AMENDMENTS	51
	PLAN TERMINATION; DISCONTINUANCE OF CONTRIBUTIONS	51
13.3		51
13.4	DISTRIBUTIONS ON PLAN TERMINATION	51
15.4	DISTRIBUTIONS ON I LAW TERMINATION	JI
ARTICLE XI	V - GENERAL PROVISIONS	52
14.1	No Contract of Employment	52
	PAYMENTS SOLELY FROM TRUST FUND	52
	INCOMPETENCY	52
	ALIENATION AND QDROs	52
	Notice to the Committee	53
14.6	Mergers and Transfers	53
	FIDUCIARIES	53
	PLANS SHALL COMPLY WITH LAW; CHOICE OF LAW	53
	ERISA 404(c)	53
	GENDER	54
	DEEMED DISTRIBUTIONS OF UNVESTED AMOUNTS	54
	HEADINGS	54
	MISSING PAYEES	54
	Changes in Vesting Schedule	54
	Tax Withholding	55
	Common Trust Funds	55
ARTICLE XV	- ESOP PROVISIONS	56
322 11		
15.1	ESOP PORTION OF THE PLAN	56
15.2	Distribution of Dividends	56
SCHEDULE A	A MINIMUM REQUIRED DISTRIBUTIONS	A-1
SCHEDULE 1	B AND APPENDICES	
	iv	L3 Technologies Master Savings Plan

INTRODUCTION

L3 Technologies, Inc. (the "Company") maintains the L3 Technologies Master Savings Plan (the "Plan") to provide retirement benefits to eligible employees of the Company, its subsidiaries and its business units.

The L3 Technologies Master Savings Plan is comprised of two parts. The first part consists of this Plan document, which sets forth the provisions that apply to all eligible employees. This restated Plan document is effective as of January 1, 2017.

The second part consists of Schedule B and Appendices, which set forth the specific benefits, rights and features that apply to employees of each of the participating business units. Each of the Appendices is effective as of the date stated therein.

The Plan was originally adopted effective May 1, 1997 by L-3 Communications Corporation. Effective after the close of business on December 31, 2016, L-3 Communications Corporation changed its name to L3 Technologies, Inc. Accordingly, the name of the Plan was changed from the L-3 Communications Corporation Master Savings Plan to the L3 Technologies Master Savings Plan effective January 1, 2017.

The Plan was amended and restated effective August 1, 2013 to include within the Plan an employee stock ownership plan, within the meaning of Section 4975(e)(7) of the Code, designed to invest primarily in L3 Stock (the "ESOP"). The ESOP will be comprised of that portion of each Participant's After-Tax Contribution Account, Catch-Up Contribution Account, Matching Contribution Account, Pre-Tax Contribution Account, Rollover Contribution Account, Roth Elective Deferral Account and Supplemental Contribution Account that is invested in the L3 Stock Fund. The L3 Stock Fund, which has previously been established as an Investment Fund shall be continued at all times as an available Investment Fund under the Plan. The Plan is a profit sharing plan, except for the ESOP portion of the Plan which is a stock bonus plan.

The Plan is intended to be qualified under Section 401(a) of the Internal Revenue Code of 1986, as amended (the "Code") and its Trust is intended to be tax-exempt under Code Section 501(a). Participants are entitled to receive benefits in accordance with the terms of the Plan in effect on the date they terminate employment or retire.

L3 Technologies Master Savings Plan

ARTICLE I - DEFINITIONS

As used in this Plan, the following terms shall have the meanings set forth herein.

1.1 Account

"Account" means the individual account or accounts established for a Participant to record contributions as adjusted for gains, including earnings, and losses

1.2 Affiliate

"Affiliate" means the Company and any entity which is required to be aggregated with the Company for purposes of the controlled group rules of Code Section 414(b), the common control rules of Code Section 414(c), the affiliated service group rules of Code Section 414(m), the rules of Code Section 414(o), and, solely for purposes of applying the rules under Section 5.5, the rules of Code Section 415(h).

1.3 After-Tax Contributions

"After-Tax Contributions" means contributions made pursuant to Section 4.2 of the Plan by a Participant on an after-tax basis.

1.4 AFTER-TAX CONTRIBUTION ACCOUNT

"After-Tax Contribution Account" means the Account established for a Participant to record After-Tax Contributions and after-tax contributions made under the terms of a Prior Plan and transferred to this Plan, as adjusted for gains, including earnings, and losses.

1.5 Beneficiary

"Beneficiary" means the Participant's beneficiary, as designated by the Participant by providing a designation to the Recordkeeper. Such designation may be revoked or changed by providing notice to the Recordkeeper. A designation or change of beneficiary designation shall be delivered to the Recordkeeper in accordance with the Plan's written administrative procedures. If upon the death of the Participant there is no properly designated beneficiary then living, "Beneficiary" shall mean the first surviving class of the following classes of beneficiaries: (a) the Participant's surviving spouse, (b) the Participant's surviving children per stirpes (excluding stepchildren but including adopted children), and (c) the Participant's estate. Notwithstanding the foregoing, for a Participant who is legally married, "Beneficiary" shall be the Participant's legal spouse at the time of death unless the Participant designates another beneficiary with the written consent of the Participant's spouse, which consent acknowledges the specific non-spouse beneficiary, and is given in accordance with the provisions of the Code. Such consent shall not be valid if the Participant subsequently changes his or her beneficiary designation unless the consent form states that the Participant may subsequently change the beneficiary. As required by the context of the Plan, the term "Beneficiary" shall include alternate payees, as defined in Code Section 414(p). If a married Participant designated his or her spouse as Beneficiary and the Plan is provided with written proof of a subsequent legal divorce with such spouse, his or her exspouse shall be deemed to have predeceased the Participant for purposes of this Beneficiary designation except to the extent an applicable court order provides that death benefits are payable to the ex-spouse.

1.6 Board of Directors

"Board of Directors" means the Board of Directors of the Company.

1.7 CATCH-UP CONTRIBUTIONS

"Catch-Up Contributions" means the contributions made by an Employer at the election of a Participant under Section 4.3, which contributions meet the requirements of, and are described in, Section 414(v) of the Code. Such Catch-Up Contributions shall not be taken into account for purposes of the Plan provisions implementing the limitations of Sections 402(g) and 415 of the Code. The Plan shall not be treated as failing to satisfy the Plan provisions implementing the requirements of Section 401(a)(4), 401(k)(3), 401(k)(11), 410(b) or 416 of the Code, as applicable, by reason of the making of such Catch-Up Contributions.

1.8 <u>Catch-Up Contribution Account</u>

"Catch-Up Contribution Account" means the Account established for a Participant to record Catch-Up Contributions and catch-up contributions made under the terms of a Prior Plan and transferred to this Plan, as adjusted for gains, including earnings, and losses.

1.9 Code

"Code" means the Internal Revenue Code of 1986, as amended from time to time, and all appropriate regulations and administrative guidance.

1.10 Committee

"Committee" means the Benefit Plan Committee, which administers the Plan.

1.11 Company

"Company" means L3 Technologies, Inc., a Delaware corporation. The Company shall act by resolution of its Board of Directors.

1.12 Compensation

"Compensation" has the meaning provided in the applicable Appendix and includes Basic Compensation, Compensation in any Plan Year shall be taken into account up to, but shall not exceed, the limit in Section 401(a)(17) of the Code in effect for that Plan Year. Any increase in the Section 401(a)(17) limit shall not apply to years preceding the first year for which the increase is effective. If a cost of living adjustment is declared under the Code Section 401(a)(17) with respect to any calendar year, it shall affect the Compensation for the Plan Year that begins on the January 1st of that same calendar year.

Solely for purposes of the nondiscrimination rules and top-heavy rules of Sections 5.3, 5.4 and 11.2, Compensation shall mean as defined in Section 5.5(d). For purposes of determining the amount of any contributions to this Plan, Compensation shall not include stock-based compensation (whether settled in cash or stock) or discretionary cash appreciation awards. For the avoidance of doubt, stock-based compensation includes cash payments made as dividend equivalents on stock-based awards, and cash payments earned as a result of stock price appreciation or total shareholder returns (whether on a relative or absolute basis).

For any Employer listed on Schedule B, Compensation shall not be determined in accordance with any Appendix, but instead shall be determined in accordance with the following paragraphs, subject to the limit set forth in Section 401(a)(17) of the Code in effect for that Plan Year and such other rules under Code Section 401(a)(17) as are set forth in the foregoing provisions of this Section:

Compensation shall mean all cash remuneration that is paid to the Participant by his or her Employer during the Plan Year and includible in gross income, including regular earnings; commissions; overtime compensation; performance based bonuses; regular vacation pay; and elective payroll deduction contributions under Code Sections 125, 132(f)(4) and 401(k).

Compensation shall include performance based bonuses for Participants employed by an Employer only if so provided on Schedule B with respect to that Employer.

Compensation shall not include non-performance based bonuses; incentive pay; severance payments; termination incentive payments; lump sum vacation allowances; taxable fringe benefits; stock-based compensation (whether settled in cash or stock); imputed income from life insurance; employer contributions to any qualified retirement plan, nonqualified deferred compensation plan or welfare plan; employee deferrals or contributions to any nonqualified deferred compensation plan; distributions from any qualified retirement plan, nonqualified deferred compensation plan or welfare plan; or any reimbursed expenses, such as relocation expenses and education expenses; and any other item not specifically included in Compensation herein.

1.13 <u>Dividend Election</u>

"Dividend Election" means the election of a Participant or Beneficiary in accordance with Section 15.2 that cash dividends on L3 Stock paid to the Participant's or Beneficiary's ESOP Account are either: (a) distributed in cash to the Participant or Beneficiary no later than 90 days after the last day of the Plan Year in which the dividend is paid; or (b) reinvested in the L3 Stock Fund allocated to the Participant's or Beneficiary's ESOP Account.

1.14 ELIGIBLE EMPLOYEE

"Eligible Employee" means an Employee who meets the requirements of the applicable Appendix.

1.15 Employee

"Employee" means any person who is a common-law employee of the Employer, but excluding any individual who is (a) an independent contractor, (b) a person included in a unit of employees covered by a collective bargaining agreement which does not expressly provide for such person's participation in the Plan, (c) an employee with no U.S. source income, or (d) a "leased employee" within the meaning of Code Section 414(n). The Employer's classification of a person at the time services are performed by such person shall be conclusive. No reclassification of a person's status with an Employer, for any reason, without reason to whether it is initiated by a court, governmental agency or otherwise and without regard to whether or not the Employer agrees to such reclassification, either retroactively or prospectively, shall result in the person being regarded as an Employee during such time. As used in each Appendix, Employee means the Employee of the Employer named in the Appendix.

1.16 Employee Contributions

"Employee Contributions" means a Participant's Pre-Tax Contributions, After-Tax Contributions, Catch-Up Contributions and Rollover Contributions.

1.17 Employee Contribution Account

"Employee Contribution Account," means the Pre-Tax Contribution Account, After-Tax Contribution Account, Catch-Up Contribution Account, Rollover Contribution Account and Roth Elective Deferral Account.

1.18 EMPLOYER

"Employer" means each business unit that participates in the Plan in accordance with the terms of an applicable Appendix.

1.19 EMPLOYER CONTRIBUTION ACCOUNT

"Employer Contribution Account" means the Matching Contribution Account, Supplemental Contribution Account and any other Account maintained by the Recordkeeper to record employer contributions.

1.20 <u>Employer Contributions</u>

"Employer Contributions" means the Matching Contributions, Supplemental Contributions and any other contributions made by the Employer pursuant to the terms of an applicable Appendix or made under the terms of a Prior Plan and transferred to this Plan.

5

1.21 <u>ERISA</u>

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and all appropriate regulations and administrative guidance.

1.22 <u>ESOP</u>

"ESOP" means the employee stock ownership plan portion of the Plan.

1.23 ESOP Account

"ESOP Account" means the Account maintained for a Participant in the ESOP that is comprised of that portion of a Participant's After-Tax Contribution Account, Catch-Up Contribution Account, Matching Contribution Account, Pre-Tax Contribution Account, Rollover Contribution Account, Roth Elective Deferral Account and Supplemental Contribution Account that is invested in the L3 Stock Fund.

1.24 FORMER PARTICIPANT

"Former Participant" means an individual who is not an Employee but has an account balance under the Plan.

1.25 HIGHLY COMPENSATED EMPLOYEE

"Highly Compensated Employee" means an Employee who is employed by the Company during the determination year and is:

- (a) An Employee who was a five-percent owner (as defined in Section 416(i)(1) of the Code) at any time during the determination year or the look-back year; or
- (b) An Employee who received compensation in excess of \$80,000 (or such higher amount as may be established from time to time by the Internal Revenue Service) and was in the top-paid group for the look-back year.

For purposes of the definition of "Highly Compensated Employee": the "determination year" is the Plan Year for which the determination of who is a highly compensated employee is being made; the "look back year" is the Plan Year immediately preceding the determination year; the "top-paid group" is the group consisting of the top 20 percent of Employees when ranked on the basis of compensation paid during the look-back year; "compensation" is defined within the meaning of Section 415(c)(3) of the Code and includes elective or salary reduction contributions to a cafeteria plan or a cash or deferred arrangement; and employers aggregated under Section 414(b), (c), (m) or (o) are treated as a single employer.

For purposes of determining the number of Employees in the top-paid group: Employees who normally work less than 17.5 hours per week and Employees who normally work not more than six months during any year shall not be excluded; and Employees who have not completed six months of service and Employees who have not attained age 21 shall be excluded.

1.26 Hour of Service

"Hour of Service" means

- (a) Each hour for which the Employee is paid, or entitled to payment, directly or indirectly, from an Employer or an Affiliate.
- (b) Each hour for which back pay, irrespective of mitigation of damages, is awarded to the Employee or agreed to by an Employer or an Affiliate
- Each hour for which an Employee is paid or entitled to payment by an Employer or an Affiliate on account of a period of time during which no duties are performed due to vacation, holiday, illness, incapacity (including disability), lay-off, jury duty, military duty or leave of absence. An Hour of Service for which an Employee is directly or indirectly paid, or entitled to payment, on account of a period during which the Employee performed no duties shall not be credited to the Employee if such payment is made or due under a plan maintained solely for the purpose of complying with any applicable worker's compensation, disability insurance, or unemployment compensation law. Hours of Service also shall not be credited for a payment which solely reimburses the Employee for medical or medically related expenses incurred by the Employee. Not more than 501 Hours of Service shall be credited under this subsection to the Employee on account of any single continuous period during which the Employee performs no duties (whether or not such period occurs in a single Computation Period).
- (d) Hours of Service performed for the Employer as a "leased employee," as defined in Section 414(n) of the Code, shall be taken into account for eligibility and vesting purposes only.
- (e) Solely for purposes of determining whether an Employee has incurred a Break-in-Service, an Employee who is not otherwise credited with an Hour of Service under subsection (a), (b) or (c) above, shall be credited with an Hour of Service for each additional hour which is part of an Employee's customary work week with an Employer or an Affiliate during which the Employee is on an unpaid authorized leave of absence, provided the Employee resumes employment with an Employer or an Affiliate upon the expiration of such authorized leave of absence.
- (f) Solely for purposes of determining whether an Employee has incurred a Break-in-Service, an Employee who is absent from work for maternity or paternity reasons and who is not otherwise credited with an Hour of Service under subsection (a), (b), (c) or (d), above, shall receive credit for the Hours of Service for which he would have been regularly scheduled had the Employee performed duties for an Employer or an Affiliate during such absence. For purposes of such determination, an absence from work for maternity or paternity reasons means an absence (1) by reason of the pregnancy of the Employee, (2) by reason of the birth of a child of such Employee, (3) by reason of the placement of a child with the Employee in connection with the adoption of such child by the Employee, or (4) for purposes of caring for such child for a period beginning immediately following such birth or placement. Hours of Service credited for purposes of such determination shall be credited in the Plan Year in which such absence begins, if necessary to prevent a One Year Period of Severance, or, in all other cases, in the next following Plan Year. In no event will more than 501 Hours of Service be credited for any single continuous period of time during which the person did not or would not have performed duties.

(g) The same Hours of Service shall not be credited more than once under subsections (a), (b), (c) or (d) above. The determination of Hours of Service for reasons other than the performance of duties shall be made in accordance with the provisions of Labor Department Regulations, 29 C.F.R. § 2530.200b-2(b), and Hours of Service shall be credited to computation periods in accordance with the provisions of Labor Department Regulations, 29 C.F.R. § 2530.200b-2(c).

1.27 <u>Investment Fund</u>

"Investment Fund" means the investment funds offered under the Plan. Except for the L3 Stock Fund, the Investment Funds may be changed by the Committee from time to time without formal plan amendment.

1.28 <u>L3 Sтоск</u>

"L3 Stock" means, with respect to periods beginning on or after January 1, 2017, the common stock of L3 Technologies, Inc. and with respect to periods ending on or before December 31, 2016, the common stock of L-3 Communications Holdings, Inc.

1.29 L3 Stock Fund

"L3 Stock Fund" means the Investment Fund that consists of L3 Stock.

1.30 MATCHING CONTRIBUTIONS

"Matching Contributions" means the Employer contributions described in Section 4.4 of the Code.

1.31 MATCHING CONTRIBUTION ACCOUNT

"Matching Contribution Account" means the Account established for a Participant to record Matching Contributions and matching contributions made under the terms of a Prior Plan and transferred to this Plan, as adjusted for gains, including earnings, and losses.

L3 Technologies Master Savings Plan

1.32 <u>Non-Covered Employer</u>

"Non-Covered Employer" means a business unit of the Company or an Affiliate that is not an Employer.

1.33 Non-Covered Status

"Non-Covered Status" means a Participant's change of employment status while remaining an employee of an Employer or an Affiliate such that he is no longer an Employee as defined in the Plan.

1.34 Non-Highly Compensated Employee

"Non-Highly Compensated Employee" means any Employee or former Employee who is not a Highly Compensated Employee.

1.35 Normal Retirement Date

"Normal Retirement Date" means the Participant's 65th birthday.

1.36 PARTICIPANT

"Participant" means an Employee who has met the eligibility requirements of the applicable Appendix and, to the extent required under Section 3.2 of the Plan, has elected to participate in the Plan.

1.37 Period of Service

"Period of Service" is defined in Section 3.7(a).

1.38 Period of Severance

"Period of Severance" is defined in Section 3.7(c).

1.39 Plan

"Plan" means this L3 Technologies Master Savings Plan, as amended from time to time, which includes the Plan document, Appendices and the Trust Agreement.

1.40 PLAN YEAR

"Plan Year" means the calendar year.

1.41 PRE-TAX CONTRIBUTIONS

"Pre-Tax Contributions" means a Participant's elective deferrals made as described in Section 402(g)(3) of the Code and made pursuant to Section 4.1 of the Plan.

1.42 Pre-Tax Contribution Account

"Pre-Tax Contribution Account" means the Account established for a Participant to record Pre-Tax Contributions and pre-tax contributions made under the terms of a Prior Plan and transferred to this Plan, as adjusted for gains, including earnings, and losses.

1.43 PRIOR PLAN

"Prior Plan" means a tax-qualified plan maintained by a prior employer.

1.44 Recordkeeper

"Recordkeeper" means the third party recordkeeper for the Plan.

1.45 ROLLOVER CONTRIBUTIONS

"Rollover Contributions" means the contributions made by an Employee in cash of an amount described in and subject to the provisions of Sections 401(a)(31), 402, 403 or 408 of the Code.

1.46 ROLLOVER CONTRIBUTION ACCOUNT

"Rollover Contribution Account" means the Account established for a Participant to record Rollover Contributions and rollover contributions made under the terms of a Prior Plan and transferred to this Plan, as adjusted for gains, including earnings, and losses.

1.47 Roth Elective Deferral Account

"Roth Elective Deferral Account" means the Account established for a Participant to record "designated Roth contributions," as defined in Code Section 402A, made under the terms of a Prior Plan and transferred to this Plan, as adjusted for gains, including earnings, and losses. No new designated Roth contributions may be made under this Plan.

1.48 Service

"Service" means the period for which an Employee is paid or is entitled to payment, subject to the rules and restrictions of Article III, for the performance of duties for an Employer, and, solely for purposes of eligibility and vesting, for the Company and an Affiliate. If an Employee transfers employment to the Company or an Affiliate in connection with a corporate acquisition, Service includes an Employee's period of employment with the prior employer.

1.49 SEVERANCE FROM SERVICE DATE

"Severance From Service Date" is defined in Section 3.7(b).

1.50 Supplemental Contributions

"Supplemental Contributions" means the contributions, other than Matching Contributions, made by an Employer pursuant to an applicable Appendix, which contributions may be referred to in an Appendix as "supplemental contributions" or "profit sharing contributions" or "employer contributions".

1.51 Supplemental Contribution Account

"Supplemental Contribution Account" means the Account established for a Participant to record Supplemental Contributions and supplemental contributions made under the terms of a Prior Plan and transferred to this Plan, as adjusted for gains, including earnings, and losses.

1.52 <u>Termination of Employment</u>

"Termination of Employment" means a severance from employment within the meaning of Section 401(k)(2)(B) of the Code.

1.53 Total Disability

"Total Disability" means a Participant is considered to be totally and permanently disabled as determined under the Employer's administrative and payroll procedures.

1.54 Trust or Trust Fund

"Trust" or "Trust Fund" means the fund held by the Trustee under the Trust Agreement to which contributions to the Plan shall be made and out of which withdrawals and distributions under the Plan shall be paid.

1.55 Trust Agreement

"Trust Agreement" means the Trust Agreement between the Company and the Trustee, pursuant to which the Plan is funded, as in effect from time to time. The Trust Agreement is incorporated by reference into, and is fully a part of, the Plan.

1.56 Trustee

"Trustee" means the trustee at any time acting under the Trust Agreement.

1.57 <u>Valuation Date</u>

"Valuation Date" means the end of each business day.

ARTICLE II - ADMINISTRATION

2.1 Committee

The Committee shall consist of members appointed by the Board of Directors to serve at its pleasure. Any member of the Committee may resign by delivering his written resignation to the General Counsel of the Company.

2.2 <u>Discretionary Power to Interpret and Administer the Plan</u>

- (a) Subject to the limitations of the Plan, the Committee shall establish rules for the administration and interpretation of the Plan. The determination of the Committee as to any disputed question shall be conclusive. All actions, decisions and interpretations of the Committee in administering the Plan shall be performed in a uniform and non-discriminatory manner.
- (b) The Committee has complete discretionary and final authority to determine all questions concerning the interpretation and administration of the Plan. The administrative decisions and Plan interpretations made by the Committee shall be given full deference by any court of law.
- (c) Each member of the Committee may delegate committee responsibilities among the Company's directors, officers or employees, and may consult with and hire outside experts.
- (d) Employees of the Company or an Affiliate who are human resources personnel or benefits representatives shall, under the authority of the Committee, perform the routine administration of the Plan, such as distributing and collecting forms, creating rules and procedures, and providing information about Plan procedures.
- (e) Should any individual receive oral or written information concerning the Plan, which is contradicted by a subsequent determination by the Committee, the Committee's final determination shall control.

2.3 General Provisions

- (a) The members of the Committee may authorize one or more of their members to execute or deliver any instrument, make any payment or perform any other act which the Plan authorizes or requires the Committee to do.
- (b) Any act which the Plan authorizes or requires the Committee to do must be done by a majority of its members. The action of such majority shall constitute the action of the Committee and shall have the same effect for all purposes as if assented to by all members of the Committee at the time in office.
- (c) The Committee may employ counsel and other agents and may procure such clerical, accounting and other services as they may require in carrying out the provisions of the Plan.

- (d) No member of the Committee shall receive any compensation for his or her services as such.
- (e) All expenses of administering the Plan, including, but not limited to, fees of accountants and counsel, shall be paid from the Trust Fund except to the extent paid by the Company.
- (f) For purposes of ERISA, the Company shall be the "Named Fiduciary" and the "Plan Administrator" and is hereby designated as agent for service of legal process for the Plan. The Company may delegate any and all of its responsibilities as Named Fiduciary and as Plan Administrator among its directors, officers or employees and may consult with and hire outside experts.

2.4 POWER TO EXECUTE PLAN AND GOVERNMENT DOCUMENTS

Any appointed Vice President of the Company shall have the authority to execute governmental filings or other documents relating to the Plan, or the Company, through action of its Board of Directors, may delegate this authority to another officer or employee of the Company.

2.5 Claims Procedure

- (a) The Committee shall make all determinations as to the right of any person to benefits. The Committee shall adopt procedures for the presentation of claims for benefits and for the review of the denial of such claims by the Committee. The decision of the Committee upon such review shall be final, subject to appeal rights provided by law.
- (b) Any legal action for benefits under the Plan must be commenced within two years of the date that an initial claim for benefits was filed with the Plan Administrator. The Plan Administrator will be the necessary party to any action or proceeding involving the assets held with respect to the Plan or the administration thereof. No Employee, Participant, Former Participant or their Beneficiaries, or any other person having or claiming to have an interest in the Plan will be entitled to any notice or process. Any final judgment that may be entered in any such action or proceeding will be binding and conclusive on all persons having or claiming to have any interest in the Plan.

2.6 <u>Indemnification</u>

To the fullest extent permitted by law, the Company agrees to indemnify, to defend, and hold harmless the members of the Committee, individually and collectively, against any liability whatsoever for any action taken or omitted by them in good faith in connection with the Plan or their duties hereunder and for any expenses or losses for which they may become liable as a result of any such actions or non-actions unless resultant from their own gross negligence or willful misconduct as determined by the Board of Directors, and the Company shall purchase insurance for the Committee to cover any of their potential liabilities with regard to the Plan and Trust.

ARTICLE III - PARTICIPATION

3.1 General Conditions of Eligibility

- (a) Each Employee shall be eligible to participate in the Plan after satisfying the eligibility requirements set forth in the applicable Appendix. For any Employer listed on Schedule B, eligibility shall not be determined in accordance with any Appendix, but instead each Employee of such Employer shall be eligible to participate in the Plan on the date he or she completes one Hour of Service.
- (b) An Employee shall be eligible to participate in the Plan only during those periods during which the Employee is in the Service of an Employer.
- (c) A Participant who remains employed with an Employer, but who ceases to be an Employee because of a change in employment status shall become a Former Participant. Accounts of all Former Participants shall (unless liquidated) continue to be adjusted by other amounts properly credited or debited to such Accounts pursuant to Article VI of the Plan.

3.2 <u>Election to Participate</u>

Each Employee who satisfies the eligibility requirements of the applicable Appendix shall become a Participant only upon making proper application in accordance with procedures established by the Committee and the Recordkeeper; provided, however, that if the Employer makes Employer Contributions that are not conditioned on the Employee's election to make Pre-Tax Contributions or After-Tax Contributions, the Employee shall automatically become a Participant, without making an application, upon satisfying the eligibility requirements for such Employer Contributions.

3.3 Transfers from Non-Covered Status

In the event an individual who is in Non-Covered Status or is an employee of a Non-Covered Employer is transferred to and becomes an Eligible Employee, such individual shall be eligible to participate in the Plan in accordance with the terms of the applicable Appendix, and his or her Service shall include his or her Service in Non-Covered Status or with the Non-Covered Employer; provided, however, that if he was eligible to participate in the Plan or any qualified defined contribution plan maintained by the Company or an Affiliate immediately prior to such transfer, he shall be eligible to participate in the Plan on the first day of the payroll period next following the date he becomes an Employee hereunder.

3.4 Transfer to Non-Covered Status

In the event a Participant is transferred to Non-Covered Status or to a Non-Covered Employer, no further contributions shall be made on behalf of the Participant. The Participant shall continue to receive vesting credit under the Plan for service in Non-Covered Status or with a Non-Covered Employer.

3.5 Transfers Among Participating Employers

- (a) If an Employee transfers employment from one Employer to another Employer after he or she is eligible to participate in the Plan, the Employee shall continue to be eligible to participate in the Plan under the terms of the Appendix that applies to the subsequent Employer, notwithstanding that the Employee may not have met the eligibility requirements of such Appendix. The Employee's account balance attributable to service with the Employer from which the Employee transferred shall continue to vest in accordance with the vesting schedule applicable to such Employer.
- (b) If an Employee transfers employment from one Employer to another Employer before he or she is eligible to participate in the Plan, the Employee shall become eligible to participate in the Plan under the terms of the Appendix that is applicable to the Employer to whom the Employee transferred, including any eligibility requirements.

3.6 ELIGIBILITY UPON RE-EMPLOYMENT

- (a) Any Employee who terminates employment with an Employer before he or she is eligible to participate in the Plan and is reemployed by any Employer shall be eligible to participate in the Plan after satisfying the eligibility requirements set forth in the applicable Appendix. For purposes of this subsection (a), all Service with the Employer, the Company and its Affiliates shall be taken into account except as otherwise provided under Section 3.7(d).
- (b) An Employee who terminates employment with the Employer after he or she is eligible to participate in the Plan and is reemployed by any Employer shall again become a Participant as of the date on which he or she again becomes an Employee, provided he or she makes proper application, if required under Section 3.2.

3.7 Service Under Elapsed Time Method

- (a) A Period of Service begins on the date the Employee first completes an Hour of Service or the date on which the Employee completes an Hour of Service following a Period of Severance and ends on his or her Severance from Service Date.
- (b) Service shall not be credited on or after any Severance from Service Date. As of a Severance from Service Date, the Participant shall become a Former Participant. A Severance from Service Date is the earlier of (1) the date on which the Employee quits, retires, is discharged or dies, or (2) the first anniversary of the first date of a period in which the Employee remains absent from Service with the Employer for any reason other than quit, retirement, discharge or death. If a quit, retirement, death, or termination occurs following an absence for any other reason (such as leave, or temporary lay-off with recall rights), but before a Period of Severance has occurred, then a Severance from Service Date will occur as of the quit, retirement, death or termination. Notwithstanding the preceding, if an Employee quits, retires, or terminates, and returns to active employment within 12 months of his initial Severance from Service Date, then his entire Period of Severance will be credited as a Period of Service for eligibility and vesting purposes, although not for contribution purposes.

- (c) A Period of Severance is the time between the Employee's Severance from Service Date and the date the Employee again performs an Hour of Service with the Employer or an Affiliate. If an Employee's absence is due to maternity or paternity leave, a Period of Severance shall begin on the first anniversary of the Employee's Severance from Service Date. A maternity or paternity leave of absence means an absence from work for any period by reason of the Employee's pregnancy, birth of the Employee's child, placement of a child with the Employee in connection with the adoption of such child, or any absence for the purpose of caring for such child for a period immediately following such birth or placement.
- (d) Should an Employee who is not vested in all his Accounts incur a Period of Severance, and again become an Employee, the Periods of Service earned before and after the Period of Severance shall be aggregated.

3.8 Qualified Military Service

Notwithstanding any provision of this Plan to the contrary, contributions, benefits and service credit with respect to "qualified military service," as defined in Code Section 414(u)(5), will be provided in accordance with Code Section 414(u). In the case of a Participant who dies while performing qualified military service, the survivors of the Participant are entitled to any additional benefits because of death, including vesting and survivor benefits contingent on termination of employment, that would have been provided under the Plan had the Participant resumed employment and then terminated employment on account of death. An individual receiving a "differential wage payment," as defined in Section 3401(h) of the Code, is treated as an employee of the Employer making the payment and the differential wage payment is treated as compensation for purposes of Code requirements applicable to the Plan.

3.9 <u>FMLA</u>

To the extent required by the Family Medical Leave Act of 1993, 29 U.S.C. § 2601 et al., Service shall include any period for which an Employee is regularly scheduled to work but is absent for a family or medical leave of absence.

L3 Technologies Master Savings Plan

ARTICLE IV - CONTRIBUTIONS

4.1 Pre-Tax Contributions

- (a) A Participant may elect to have Pre-Tax Contributions deducted from his or her Compensation for each pay period in an amount elected by the Participant, which may be equal to any whole percentage of the Participant's Compensation for each such pay period not to exceed 25 percent (or such other percentage as may be designated by the Company, in writing, without formal plan amendment).
- (b) A Participant may change the amount of, or suspend, his or her Pre-Tax Contributions as of any date.
- (c) A Participant who has suspended his or her Pre-Tax Contribution may resume making Pre-Tax Contributions as of any date after such suspension.
- (d) Any election described in this Section 4.1 shall be made with the Recordkeeper in accordance with the procedures established by the Committee and the Recordkeeper, and shall be effective as soon as administratively feasible after receipt by the Recordkeeper.
- (e) An Eligible Employee whose date of hire is on or after July 1, 2008 will be deemed to have elected that a Pre-Tax Contribution of three percent of the Eligible Employee's Compensation be deducted from the Eligible Employee's Compensation for each payroll period starting on or after the 60th day following such date of hire and credited to the Eligible Employee's Pre-Tax Contribution Account, subject to the following conditions and requirements:
 - Unless an Eligible Employee elects otherwise, the Pre-Tax Contributions will be invested in an investment fund designated by the Committee that satisfies the requirements for a "qualified default investment alternative" (QDIA) under regulations issued by the U.S. Department of Labor. The Eligible Employee may transfer all or a portion of the amounts invested in the QDIA to other investment alternatives under the Plan to the same extent as Participants who affirmatively elected to invest in the QDIA; provided, however, that any such transfer within the first 90 days following the initial automatic Pre-Tax Contribution shall not be subject to any restrictions, fees or expenses (including without limitation surrender charges, liquidation or exchange fees, redemption fees and similar expenses charged in connection with the liquidation of, or transfer from, the QDIA) and following such 90-day period shall not be subject to any such restrictions, fees and expenses that are not otherwise applicable to Participants who affirmatively elected to invest in the QDIA.
 - (2) An Eligible Employee described in this Section 4.1(e) may change the amount of or suspend automatic Pre-Tax Contributions at any time by making an affirmative election under Sections 4.1(b) or (c) in accordance with rules established by the Committee.

- Each Eligible Employee will be provided with a notice written in a manner calculated to be understood by the average Eligible Employee no less than 30 days and no more than 90 days prior to the Eligible Employee's first automatic Pre-Tax Contribution and no less than 30 days and no more than 90 days prior to the beginning of each subsequent Plan Year that includes the following information: the automatic Pre-Tax Contribution election that will be made on the Eligible Employee's behalf if the Eligible Employee does not make an affirmative election; the Eligible Employee's rights to suspend or change the Pre-Tax Contribution election in accordance with Sections 4.1(b) and (c); the investment of automatic Pre-Tax Contributions in the QDIA in the absence of any other investment election by the Eligible Employee; a description of the QDIA including investment objectives, risks and return characteristics and fees and expenses; the right of Eligible Employees on whose behalf assets are invested in a QDIA to direct the investment of such assets to another investment alternative under the Plan, including any restrictions, fees or expenses applicable to such transfer; an explanation of where the Eligible Employee can obtain information concerning other investment alternatives under the Plan; and such other information and as may be required by applicable law.
- (4) The Eligible Employee will receive a copy of the most recent prospectus for the QDIA provided to the Plan no later than immediately after the initial automatic Pre-Tax Contribution is invested in the QDIA. In addition, the Eligible Employee will be entitled to receive upon written request such other information as may be made available upon request to Participants who affirmatively elect that a portion of their Accounts be invested in the QDIA.
- (5) The Eligible Employee will receive any materials provided to the Plan relating to the exercise of voting, tender or similar rights with respect to the QDIA to the extent those rights are passed through to Participants under the terms of the Plan as well as a description of any Plan provisions relating to the exercise of such rights.
- (6) This Section 4.1(e) will be applicable to Eligible Employees subject to collective bargaining agreements only to the extent permitted under the terms of the applicable collective bargaining agreement.

4.2 After-Tax Contributions

A Participant may elect to have After-Tax Contributions deducted from his or her Compensation for each pay period in an amount elected by the Participant, which may be equal to any whole percentage of the Participant's Compensation for each such pay period not to exceed 25 percent (or such other percentage as may be designated by the Company, in writing, without formal plan amendment), less the amount of the Participant's Pre-Tax Contributions for such pay period. An election to make After-Tax Contributions shall be made in accordance with procedures established by the Committee and the Recordkeeper, and shall be effective as soon as administratively feasible after receipt by the Recordkeeper.

4.3 <u>Catch-Up Contributions</u>

A Participant who is at least age 50 by December 31 of a calendar year may elect to have Catch-Up Contributions deducted from his or her Compensation in an amount elected by the Participant, which may be equal to any whole percentage not to exceed 50 percent of his or her Compensation for a pay period (or such other percentage as may be designated by the Company, in writing, without formal plan amendment). An election to make Catch-Up Contributions will be subject to the rules of Section 4.1(b), (c) and (d).

4.4 <u>Matching Contributions</u>

(a) The Employer shall make Matching Contributions with respect to Pre-Tax and After-Tax Contributions made by a Participant in an amount determined under the formula set out in the applicable Appendix.

For any Employer listed on Schedule B, Matching Contributions shall not be determined in accordance with any Appendix, but instead shall be determined in accordance with the following matching contribution formula:

The Employer shall make Matching Contributions on behalf of each Participant who makes Pre-Tax Contributions or After-Tax Contributions for a payroll period in an amount equal to the percent of the Participant's aggregate Pre-Tax Contributions and After-Tax Contributions set forth on Schedule B for that Employer that do not exceed the percent of Compensation for the payroll period set forth on Schedule B for that Employer. The Employer shall make Matching Contributions on behalf of each Participant who makes Catch-up Contributions for a payroll period under the same formula.

In addition, for any Employer listed on Schedule B as providing "true-up contributions," the Employer shall make an additional Matching Contribution for a Plan Year equal to the difference between (1) and (2) where:

- (1) is the amount that would have been contributed as matching contributions for the Plan Year by the Employer if the matching contribution formula for the Employer were applied on an annual, rather than a pay period, basis; and
- (2) is the amount that would have been contributed as matching contributions for the Plan Year by the Employer if the matching contribution formula for the Employer were applied only on a pay period basis.

Notwithstanding the foregoing, if specifically provided on Schedule B, the Employer shall make Matching Contributions on behalf of each Participant who makes Pre-Tax Contributions or After-Tax Contributions for a Plan Year (instead of a payroll period) in an amount equal to the percent of the Participant's aggregate Pre-Tax Contributions and After-Tax Contributions set forth on Schedule B for that Employer that do not exceed the percent of Compensation for the Plan Year set forth on Schedule B for that Employer. The Employer shall make Matching Contributions on behalf of each Participant who makes Catch-up Contributions for a payroll period under the same formula.

- (b) The Employer shall make additional Matching Contributions with respect to Catch-up Contributions under the same formula applicable to Pre-Tax and After-Tax Contributions.
- (c) Notwithstanding any other provision of the Plan or any Appendix to the contrary, no additional Matching Contribution or additional other Employer Contribution will be made to the Plan with respect to a Participant who is treated as an "Additional SERP Participant" under the L3 Technologies, Inc. Supplemental Executive Retirement Plan.
- (d) Notwithstanding any other provision of the Plan or any Appendix to the contrary, effective as of February 15, 2012, no Matching Contributions shall be made with respect to Catch-up Contributions made by Participants that are employed by L-3 Services, Inc., International Resources Group Ltd., Engility Corp., L-3 National Security Solutions, Inc., or the Engineering and Technical Services (E&TS) business unit of the Company.

4.5 Other Employer Contributions

The Employer shall make such other Employer contributions as set forth in the applicable Appendix. For any Employer listed on Schedule B, Supplemental Contributions shall continue to be determined in accordance with the Appendix in effect on December 31, 2010 except as otherwise provided on Schedule B for that Employer.

4.6 <u>Rollover Contributions</u>

An Employee may make a Rollover Contribution at any time regardless of whether the Employee has met the eligibility requirements of the applicable Appendix and regardless of whether the Employee has elected to make Pre-Tax Contributions or After-Tax Contributions. A Rollover Contribution shall be paid to the Trustee in cash. The Committee shall develop such procedures and require such information from an individual desiring to make a Rollover Contribution as it deems necessary or desirable to determine that the proposed contribution will meet the requirements for a Rollover Contribution as set forth in the Plan and the Code, and for the return of Rollover Contributions, and the earnings and losses thereon, which have been determined to have been invalidly made.

4.7 <u>Contributions Required by the terms of a Collective Bargaining Agreement</u>

To the extent the provisions of a collective bargaining agreement provide for employer or employee contributions in an amount that is different than the amount provided under the Plan, including an applicable Appendix, the terms of the collective bargaining agreement shall apply.

4.8 Suspension of Contributions Upon Transfer to Non-Covered Status

In the event a Participant is transferred to Non-Covered Status or to a Non-Covered Employer, the Participant shall become a Former Participant and his or her Employee Contributions, if any, shall be automatically suspended as of the date of such transfer.

4.9 Timing of Contributions to Trustee

- (a) Each Employer shall pay to the Trust an amount equal to the Participants' Employee Contributions as soon as practicable after such amounts are deducted from their remuneration, but not later than required under applicable law.
- (b) Employer Contributions that are required to be made for a calendar year shall be paid into the Trust no later than the time prescribed by law for filing the Company's Federal income tax return, including extensions, for such calendar year.

4.10 METHOD BY WHICH CONTRIBUTIONS ARE MADE TO THE TRUST

Except as otherwise provided in the applicable Appendix, Employer Contributions shall be made in shares of L3 Stock. Employee Contributions shall be made in cash.

4.11 Transfers from Prior Plan

- (a) <u>Time and Manner</u>. Upon the direction of the Committee, the Trustee shall accept the assets and liabilities representing the account balances under a Prior Plan of any participant or former participant in a Prior Plan ("Prior Plan Participant").
- (b) <u>Beginning Account Balances After Transfer</u>. Absent an election from the Prior Plan Participant, amounts transferred on behalf of a Prior Plan Participant from a Prior Plan pursuant to subsection (a) above shall be allocated among such Prior Plan Participant's Accounts under this Plan in the same way that those amounts were allocated to such accounts under the Prior Plan.
- (c) <u>Investment of Transferred Amounts</u>. Until such time as the Prior Plan Participant makes a new investment election as provided under Section 7.1, all amounts transferred from the Prior Plan shall be invested in Investment Funds that have similar characteristics as the investment funds in which such transferred amounts were invested under the Prior Plan.

(d) <u>Salary Deferral Elections</u>. Until such time as the Prior Plan Participant makes a new salary deferral election as provided under Sections 4.1, 4.2, and 4.3, the salary deferral elections under the Prior Plan shall remain in effect and shall be deemed to be an election under this Plan.

4.12 Qualified Non-Elective Contributions

- (a) The Employer may make Qualified Non-Elective Contributions and Qualified Matching Contributions, as defined in subsections (b) and (c) below, on behalf of Participants who are Non-Highly Compensated Employees. The Qualified Non-Elective Contributions and Qualified Matching Contributions, if any, will be allocated to Participants who are Non-Highly Compensated Employees in accordance with Treas. Reg. 1.401(k)-2(a)(6) and 1.401(m)-2(a)(g), respectively.
- (b) "Qualified Non-Elective Contributions" shall mean contributions other than Qualified Matching Contributions, made by the Employer that are nonforfeitable when made to the Plan and are subject to the same distribution rules as Pre-Tax Contributions, provided that Qualified Non-Elective Contributions shall not be eligible for hardship withdrawals.
- (c) "Qualified Matching Contributions" means Matching Contributions that are nonforfeitable when made to the Plan and that are distributable only in accordance with the distribution provisions (other than for hardships) applicable to Pre-Tax Contributions.

L3 Technologies Master Savings Plan

ARTICLE V - LIMITATIONS ON CONTRIBUTIONS

5.1 Suspension of Contributions Upon Reaching the Savings Maximum

- (a) A Participant's Pre-Tax Contributions shall automatically be suspended when the aggregate amount of such Pre-Tax Contributions for any Plan Year equals the maximum amount permitted under Section 402(g) of the Code. A Participant whose Pre-Tax Contributions have been suspended pursuant to paragraph (a) shall automatically have Pre-Tax Contributions deducted from his or her Compensation as of the first day of the next succeeding Plan Year at the same deferral percentage as the Participant had most recently elected prior to such suspension, provided that the Participant has not made an election to make After-Tax Contributions at any time after the date that the Pre-Tax Contributions were suspended and further provided that the Participant has not made an election to make Pre-Tax Contributions in a different amount. A Participant who, after reaching the maximum amount permitted under Section 402(g) of the Code, makes an election to make After-Tax Contributions, shall not have Pre-Tax Contributions automatically deducted from his or her Compensation for the next Plan Year until such Participant makes a new salary deferral election for such next Plan Year.
- (b) A Participant's Catch-Up Contributions shall automatically be suspended when the aggregate amount of such Catch-Up Contributions for any Plan Year equals the maximum amount permitted under Section 414(v) of the Code. A Participant whose Catch-Up Contributions have been suspended must make a new election to make Catch-Up Contributions for the Plan Year.

5.2 <u>Return of Excess Deferrals</u>

If the aggregate of the Participant's Pre-Tax Contributions or Catch-Up Contributions to this Plan and any other plan to which the Participant makes elective deferrals as defined in Section 402(g)(3) of the Code for any Plan Year exceeds the maximum amount permitted under Section 402(g) or 414(v) of the Code for such Plan Year, the Participant may notify the Recordkeeper no later than the date established by the Recordkeeper of the amount of the excess deferrals to be assigned to the Plan. If there are excess deferrals that arise by taking into account only those Pre-Tax Contributions or Catch-Up Contributions to this Plan, the Participant shall be deemed to have notified the Recordkeeper of such excess deferrals. Upon receipt of such notice (or deemed notice), the Recordkeeper shall cause an amount of Pre-Tax Contributions or Catch-Up Contributions equal to the excess deferrals allocated to the Plan and the income allocable thereto to be distributed to such Participant prior to April 15 of such following Plan Year. Pre-Tax Contributions or Catch-Up Contributions for which the Employer does not makes a Matching Contribution shall be returned before Pre-Tax Contributions for which a Matching Contribution has been made. Excess deferrals to be distributed for a taxable year will be reduced by excess contributions previously distributed under Section 5.3(b) for the Plan Year beginning in such taxable year. For purposes of this Section, the term "excess deferrals" with respect to Pre-Tax Contributions means a Participant's Pre-Tax Contributions to this Plan and to a plan maintained by any other employer that, in the aggregate, exceed the maximum amount permitted under Section 402(g) of the Code. For purposes of this Section, the term "excess deferrals" with respect to Catch-Up Contributions means a Participant's Catch-Up Contributions to the Plan and to a plan maintained by any other employer that, in the aggregate, exceed the maximum amount permitted under Section 414(v) of the Code. For purposes of this section, the income allocable to "excess deferrals" is equal to the allocable gain or loss for the Plan Year to which the excess deferrals are attributable and, with respect to distributions of "excess deferrals" occurring in 2007, the allocable gain or loss for the period after the close of such Plan Year and prior to the date of distribution.

5.3 <u>Section 401(k) Limit on Pre-Tax Contributions</u>

- (a) The Committee shall determine, during and as of the end of each Plan Year, the Actual Deferral Percentages relevant for purposes of this Section based on the actual and projected rate for each Participant of his Compensation and Pre-Tax Contributions for the remainder of the Plan Year. If, based on such determination, the Committee concludes that a reduction in the Pre-Tax Contributions for any Participant is necessary or advisable in order to comply with the limitations of paragraph (1) or (2) below, it shall so notify each affected Participant. In such event, the maximum allowable Pre-Tax Contributions shall be reduced in accordance with the direction of the Committee, and the contribution election of each Participant affected by such determination shall be modified accordingly.
 - (1) The Actual Deferral Percentage (as defined below) for the group of Highly Compensated Employees is not more than the Actual Deferral Percentage for the group of Non-Highly Compensated Employees multiplied by 1.25.
 - (2) The Actual Deferral Percentage for the group of Highly Compensated Employees is not more than the Actual Deferral Percentage for the group of Non-Highly Compensated Employees multiplied by 2.0 and is not more than 2 percentage points more than the Actual Deferral Percentage for the group of Non-Highly Compensated Employees.
 - (3) For the purposes of paragraphs (1) and (2) above:
 - (A) The "Actual Deferral Percentage" for a specified group of Participants for a Plan Year shall be the average of the ratios (calculated separately for each Participant in such group and rounded to the nearest 0.01%) of
 - (i) the amount of Pre-Tax Contributions and Qualified Non-Elective Contributions on behalf of each such Participant for such Plan Year (including the amount of any Excess Deferrals distributed to a Participant), to
 - (ii) such Participant's Compensation for such Plan Year.

- (B) For the purposes of the Actual Deferral Percentage test only, "Participant" means any Employee who is eligible to participate in the Plan for part or all of the applicable Plan Year.
- (b) If the limits of Section 5.4(a) are not met and the Employer does not make Qualified Non-Elective Contributions (as defined in Section 4.12(b)) for the Plan Year, any "excess contributions" for the Plan Year shall be distributed in cash to the Highly Compensated Employees on whose behalf they were paid into the Plan, no later than two and one-half months after the end of such Plan Year, if at all possible, and in any event no later than the close of such following Plan Year. The amount distributed to any such Participant shall be increased or decreased by a pro rata share of the net income or loss attributable to such "excess contributions" as determined by the Committee in accordance with applicable regulations. If such Participant's excess contributions are invested in more than one Investment Fund, such distribution shall be made pro rata, to the extent practicable, from all such Investment Funds. For purposes of this Section 5.3(b), "excess contributions" means, with respect to any Plan Year, the excess of (1) the aggregate amount of Pre-Tax Contributions actually paid into the Plan on behalf of Highly Compensated Employees for such Plan Year, over (2) the maximum amount of such contributions permitted for such Plan Year under the limitations set forth above, determined by reducing the amount of Pre-Tax Contributions on behalf of Highly Compensated Employees in the order of their highest Actual Deferral Percentages until the requirements of Section 5.3(a) are satisfied. Excess contributions to be distributed for a taxable year will be reduced by excess deferrals previously distributed under Section 5.2 for the Plan Year beginning in such taxable year. Any Employer Matching Contributions made with respect to excess contributions shall be forfeited. Forfeitures shall be applied to reduce contributions that the Employer is required to pay into the Plan and to pay Plan expenses.
- (c) The rules of Section 401(k)(3) and Treasury Regulation Section 1.401(k)-1 are hereby incorporated by reference.
- (d) For purposes of sub-section (b) above, the income allocable to "excess contributions" is equal to the allocable gain or loss for the Plan Year to which the excess contributions are attributable and, for distributions of "excess contributions" occurring in 2006 and 2007, the allocable gain or loss for the period after the close of such Plan Year and prior to the date of distribution.

5.4 <u>Section 401(m) Limit on Matching Contributions</u>

(a) The Committee shall determine, during and as of the end of each Plan Year, the Actual Contribution Percentage relevant for purposes of this Section, based on the actual and projected rate for each Participant of his or her Compensation, Matching Contributions, and After-Tax Contributions. If, based on such determination, the Committee concludes that a reduction in Matching Contributions or After-Tax Contributions made for any Participant is necessary or advisable in order to comply with the limitations of paragraph (1) or (2) below, it shall so notify each affected Participant. In such event, the maximum allowable Matching Contributions and After-Tax Contributions shall be reduced in accordance with the direction of the Committee.

- (1) The Actual Contribution Percentage (as defined below) for the group of Highly-Compensated Employees is not more than the Actual Contribution Percentage for the group of Non-Highly Compensated Employees multiplied by 1.25.
- (2) The Actual Contribution Percentage for the group of Highly Compensated Employees is not more than the Actual Contribution Percentage for the group of Non-Highly Compensated Employees multiplied by 2.0 and is not more than 2 percentage points more than the Contribution Percentage for the group of Non-Highly Compensated Employees.
- (3) For the purposes of paragraphs (1) and (2) above:
 - (A) The "Actual Contribution Percentage" for a specified group of Participants for a Plan Year shall be the average of the ratios (calculated separately for each Participant in such group and rounded to the nearest 0.01%) of
 - (i) the amount of Matching Contributions, After-Tax Contributions and Qualified Matching Contributions on behalf of each such Participant for such Plan Year (including the amount of any Excess Deferrals distributed to a Participant), to
 - (ii) such Participant's Compensation for such Plan Year.
 - (B) For the purposes of the Actual Contribution Percentage test only, "Participant" means any Employee who is eligible to participate in the Plan for part or all of the applicable Plan Year.
- (b) If the limits of Section 5.4(a) are not met and the Employer does not make Qualified Matching Contributions (as defined in Section 4.12(c)) for the Plan Year, any "excess aggregate contributions" for the Plan Year shall be distributed in cash to the Highly Compensated Employees on whose behalf they were paid into the Plan, no later than two and one-half months after the end of such Plan Year, if at all possible, and in any event no later than the close of such following Plan Year. The amount distributed to any such Participant shall be increased or decreased by a pro rata share of the net income or loss attributable to such "excess aggregate contributions" as determined by the Committee in accordance with applicable regulations. If such Participant's excess aggregate contributions are invested in more than one Investment Fund, such distribution shall be made pro rata, to the extent practicable, from all such Investment Funds. For purposes of this Section 5.4(b), "excess aggregate contributions" means, with respect to any Plan Year, the excess of (1) the aggregate amount of Matching Contributions or After-Tax Contributions actually paid into the Plan on behalf of Highly Compensated Employees for such Plan Year, over (2) the maximum amount of such contributions permitted for such Plan Year under the limitations set forth above, determined by reducing the amount of Matching Contributions and After-Tax Contributions on behalf of Highly Compensated Employees in the order of their highest Actual Contribution Percentages until the requirements of Section 5.4(a) are satisfied.

- (c) The rules of Section 401(m)(2) and Treasury Regulation Section 1.401(m)-1 are hereby incorporated by reference.
- (d) For purposes of sub-section (b) above, the income allocable to "excess aggregate contributions" is equal to the allocable gain or loss for the Plan Year to which the excess aggregate contributions are attributable and, for distributions of "excess aggregate contributions" occurring in 2006 and 2007, the allocable gain or loss for the period after the close of such Plan Year and prior to the date of distribution.

5.5 <u>Annual Additions Limit</u>

(a) Notwithstanding any other provision of the Plan to the contrary, the maximum amount of annual additions which may be credited to a Participant's Accounts for any Plan Year shall not exceed the lesser of (1) \$40,000 as adjusted for increases in the cost-of-living in accordance with regulations prescribed by the Secretary of Treasury; provided, however, that no such increase in the maximum dollar amount shall become effective until January 1 of the applicable calendar year and shall apply beginning with the Plan Year coincident with such calendar year); or (2) 100% (or such other percentage as determined in accordance with the Code) of the Participant's Section 415 earnings (as defined in paragraph (d) of this Section) for such Plan Year. For the purpose of this paragraph, a Participant's "annual additions" for any Plan Year shall mean the sum of (A) employer contributions and forfeitures allocable to a Participant under all plans (or portions thereof) maintained by the Company or an Affiliate subject to Section 415(c) of the Code, (B) the Participant's employee contributions under all such plans (or portions thereof), and (C) amounts described in Section 419A(d)(2) of the Code (relating to post-retirement medical benefits of key employees) or allocated to a pension plan individual medical account described in Section 415(l) of the Code, to the extent includible for purposes of Section 415(c)(2) of the Code. A Participant's employee contributions shall be determined without regard to (i) any rollover contributions, (ii) any repayments of loans, or (iii) any prior distributions repaid upon the exercise of buy-back rights. Employer and employee contributions taken into account as Annual Additions shall include "excess contributions" as defined in Section 401(k)(8)(B) of the Code, "excess aggregate contributions" as defined in Section 401(m)(6)(B) of the Code, and "excess deferrals" as described in Section 402(g) of the Code, regardless of whether such amounts are distributed or forfeited (except to the extent such "excess deferrals" are distributed to the Participant before the end of the taxable year of the Participant in which such deferrals were made).

27

- (b) If, as a result of the allocation of forfeitures, a reasonable error in estimating a Participant's annual compensation, a reasonable error in determining the amount of elective deferrals for a Participant, or other facts and circumstances permitted by Commissioner of the Internal Revenue Service, the Annual Additions to the Accounts of a Participant for any Plan Year would exceed the limitations set forth in subsection (a), such Participant's Annual Additions for such Plan Year shall be reduced by the amount required to eliminate such excess in the following order:
 - (1) After-Tax Contributions for which no Matching Contributions were made;
 - (2) Pre-Tax Contributions for which no Matching Contributions were made;
 - (3) After-Tax Contributions for which Matching Contributions were made and the Matching Contributions with respect to such After-Tax Contributions;
 - (4) Pre-Tax Contributions for which Matching Contributions were made and the Matching Contributions with respect to such Pre-Tax Contributions; and
 - (5) Supplemental Contributions.

To the extent contributions on behalf of a Participant are required to be reduced in order to meet the requirements of subsection (a) above, such After-Tax Contributions shall be returned to the Participant as soon as practicable thereafter.

- (c) For the purposes of this Section, this Plan and all other defined contribution plans (as defined in Section 414(i) of the Code) maintained by the Employer, or an Affiliate (whether or not terminated) shall be treated as one defined contribution plan.
- (d) For purposes of this Section, the following shall have the meanings set forth below:
 - (1) a Participant's "Section 415 earnings" means wages, salaries, and fees for professional services and other amounts received for personal services actually rendered in the course of employment with the Employer or an Affiliate up to, but not in excess of, the limit in Section 401(a)(17) of the Code in effect for that Plan Year (as adjusted for cost of living in accordance with that Code Section) including, but not limited to, commissions paid to salesmen, compensation for services on the basis of a percentage of profits, commissions on insurance premiums, tips, bonuses, and Pre-Tax Contributions and any employee contributions made under a plan maintained by the Employer pursuant to Sections 125, 132(f)(4) or 401(k) of the Code, and excluding Employer contributions to a plan of deferred compensation which are not includible in the Participant's gross income for the taxable year in which contributed, or Employer contributions under a simplified employee pension plan to the extent such contributions are deductible by the Participant, or any distributions from a plan of deferred compensation; amounts realized from the exercise of a non-qualified stock option, when restricted stock (or property) held by the Participant either becomes freely transferable or is no longer subject to a substantial risk of forfeiture; amounts realized from the sale, exchange or other disposition of stock acquired under an incentive stock option; and other amounts which receive special tax benefits, or contributions made by the Employer (whether or not under a salary reduction agreement) towards the purchase of an annuity described in Section 403(b) of the Code (whether or not the amounts are actually excludible from the gross income of the Participant). Amounts under Section 125 of the Code shall include amounts not available to a Participant in cash in lieu of group health coverage because the Participant is unable to certify that he or she has other health coverage. An amount will be treated as an amount under Section 125 of the Code only if the Employer does not request or collect information regarding the Participant's other health coverage as part of the enrollment process for the health plan.

For purposes of the limitation under Section 415 of the Code, "Section 415 Earnings" for the limitation year shall include compensation paid by the later of 2-½ months after a Participant's severance from employment with the Company or an Affiliate or the end of the limitation year that includes the date of the Participant's severance from employment with the Company or an Affiliate, if:

- (i) the payment is regular compensation for services during the Participant's regular working hours, or compensation for services outside the Participant's regular working hours (such as overtime or shift differential), commissions, bonuses, or other similar payments, and, absent a severance from employment, the payments would have been paid to the Participant while the Participant continued in employment with the Company or an Affiliate; or
- (ii) the payment is for unused accrued bona fide sick, vacation or other leave that the Participant would have been able to use if employment had continued; or
- (iii) the payment is received by the Participant pursuant to a nonqualified unfunded deferred compensation plan and would have been paid at the same time if employment had continued, but only to the extent includible in gross income.

Any payments not described above shall not be considered compensation if paid after severance from employment, even if they are paid by the later of 2-½ months after the date of severance from employment or the end of the limitation year that includes the date of severance from employment, except: (a) payments to an individual who does not currently perform services for the Company or an Affiliate by reason of qualified military service (within the meaning of Section 414(u)(1) of the Code) to the extent these payments do not exceed the amounts the individual would have received if the individual had continued to perform services for the Company or an Affiliate rather than entering qualified military service; or (b) compensation paid to a Participant who is permanently and totally disabled, as defined in Section 22(e)(3) of the Code, provided salary continuation applies to all Participants who are permanently and totally disabled for a fixed or determinable period, or the Participant was not a highly compensated employee, as defined in Section 414(q) of the Code, immediately before becoming disabled.

- (2) "Qualified Nonelective Contributions" means contributions other than Matching Contributions or Qualified Matching Contributions) made by the Employer and allocated to participants' accounts that the participants may not elect to receive in cash until distributed from the Plan; that are nonforfeitable when made to the Plan; and that are distributable only in accordance with the distribution provisions (other than for hardships) applicable to Pre-Tax Contributions.
- (3) "Qualified Matching Contributions" means Matching Contributions that are nonforfeitable when made to the Plan and that are distributable only in accordance with the distribution provisions (other than for hardships) applicable to Pre-Tax Contributions.

ARTICLE VI - PARTICIPANTS' ACCOUNTS

6.1 ESTABLISHMENT OF ACCOUNTS

The Committee shall establish and maintain the following Accounts for each Participant to the extent applicable:

- (a) An After-Tax Contribution Account for each Participant who makes After-Tax Contributions;
- (b) A Pre-Tax Contribution Account for each Participant who makes Pre-Tax Contributions;
- (c) A Catch-Up Contribution Account for each Participant who makes Catch-Up Contributions;
- (d) A Matching Contribution Account for each Participant for whom Matching Contributions are made;
- (e) A Supplemental Contribution Account for each Participant for whom Supplemental Contributions are made;
- (f) A Rollover Contribution Account for each Participant who makes Rollover Contributions;
- (g) A Roth Elective Deferral Account for each Participant who made "designated Roth contributions," as defined in Code Section 402A under the terms of a Prior Plan; and
- (h) Such other Accounts as may be necessary to record any additional types of contributions made for a Participant in accordance with the applicable Appendix.

6.2 <u>Accounts In Investment Funds</u>

A Participant's Accounts shall be invested in the applicable Investment Funds in accordance with the provisions of Article VII.

6.3 How Accounts are Valued

- (a) The value of a Participant's Accounts shall be determined as of the close of each Valuation Date.
- (b) The value of a Participant's Account as of any Valuation Date shall first be decreased by any withdrawals, loans or distributions from the Account made on such Valuation Date and then increased or decreased by the Account's pro rata share of income, expense, gains for such Valuation Date.

ARTICLE VII - INVESTMENT OF CONTRIBUTIONS; TRANSFERS BETWEEN FUNDS

7.1 Participant Directed Investments

- (a) A Participant shall have the right to direct the investment of Employee Contribution to be made on his or her behalf in one or more of the Investment Funds (excluding a Prior Company Stock Fund) in multiples of 1% (or such greater percentage as determined by the Committee), provided that, if no election is in effect, or made, such amounts shall be invested in a fund that has as its objective the preservation of capital, as determined by the Committee. An investment election with respect to Employee Contributions will be effective for all Employee Contributions made after the date of the election and will remain in effect until the Participant files a new investment election. A Participant may, at any time, elect to transfer part or all of the value of his or her Employee Contribution Account balance among the Investment Funds (excluding a Prior Company Stock Fund) in multiples of 1% (or such greater percentage as determined by the Committee).
- (b) With respect to Employer Contributions that are made in L3 Stock pursuant to Section 4.9, a Participant shall have the right to transfer part or all of the his or her Employer Contribution Account balance attributable to such Employee Contributions in one or more of the Investment Funds (excluding a Prior Company Stock Fund) in multiples of 1% (or such greater percentage as determined by the Committee). An investment election with respect to Employer Contributions will be effective for the Employer Contributions credited to the Participant's Employer Contributions Account on the date the election is made. Employer Contributions that are made in L3 Stock after the date of the election will remain invested in the L3 Stock Fund until the Participant makes an election to transfer such Employer Contributions out of the L3 Stock Fund.
- (c) With respect to Employer Contributions that are not made in L3 Stock pursuant to Section 4.9, a Participant shall have the right to direct the investment of such Employer Contributions in one or more of the Investment Funds (excluding a Prior Company Stock Fund) in multiples of 1% (or such greater percentage as determined by the Committee), provided that, if no election is in effect, or made, such amounts shall be invested in a fund that has as its objective the preservation of capital, as determined by the Committee. An investment election with respect to Employer Contributions will be effective for all of such Employer Contributions made after the date of the election and will remain in effect until the Participant files a new investment election. A Participant may, at any time, elect to transfer part or all of the value of his or her Employer Contribution Account balance among the Investment Funds (excluding a Prior Company Stock Fund) in multiples of 1% (or such greater percentage as determined by the Committee).
- (d) A Participant shall have the right to direct the investment of his or her Rollover Contributions in one or more of the Investment Funds (excluding a Prior Company Stock Fund) in multiples of 1% (or such greater percentage as determined by the Committee), provided that, if no election is in effect, or made, such amounts shall be invested in a fund that has as its objective the preservation of capital, as determined by the Committee. A separate election must be made for each Rollover Contribution. A Participant may, at any time, elect to transfer part or all of the value of his or her Rollover Contributions Account balance among the Investment Funds (excluding a Prior Company Stock Fund) in multiples of 1% (or such greater percentage as determined by the Committee).

(e) Investment elections shall be made electronically with the Recordkeeper in accordance with the procedures established by the Committee and Recordkeeper, and shall be effective as soon as administratively feasible after receipt by the Recordkeeper; provided, however, that the initial investment election for a Rollover Contribution shall be made in writing.

7.2 <u>Discontinued Funds</u>

In the event any existing Investment Fund is discontinued (the "Discontinued Fund"), the Committee shall provide each Participant with notice of such discontinuance. The Committee shall also provide each Participant whose Accounts are invested in the Discontinued Fund with an election period of at least 30 days in which to elect to transfer the value of his Accounts invested in the Discontinued Fund to any other Investment Funds. In the event any such Participant fails to file a timely election with respect to such transfer, the Committee shall direct the Trustee to transfer the value of such Participant's Accounts invested in the Discontinued Fund to a fixed-income Fund whose principal is not subject to decrease in value.

7.3 <u>Limitation or Suspension of Transaction and Limitation of Daily Securities Trading</u>

Except with respect to transfers into the L3 Stock Fund and notwithstanding any other provision of this Article VII, the Company shall, in its sole discretion, limit or suspend any or all Investment Fund transfers, withdrawals, distributions and loans, including subsequent Investment Fund transfers, withdrawals, distributions and loans elected prior to the determination of such limitation or suspension, in the event the Company determines, in its sole discretion, that such action is in the best interest of the Plan or the Participants. The Trustee, or the Investment Manager for a specific Investment Fund, may, in its sole discretion, limit the daily volume of its purchases or sales of securities for the Trust, other than purchases of L3 Stock.

ARTICLE VIII - VESTING

8.1 Full Vesting in Employee Contribution Account

A Participant shall always be 100 percent vested in his or her Employee Contribution Account.

8.2 Vesting in Employer Contribution Account

A Participant shall become vested in his or her Employer Contribution Account in accordance with the applicable Appendix, provided, however, that the Participant shall become 100 percent vested in such Accounts on the earlier of (1) his or her Normal Retirement Date, if the Participant is actively employed by the Employer (or an Affiliate) on that date, or (2) the date he or she terminates employment with the Employer (or an Affiliate) due to death or Total Disability.

For Participants employed by an Employer listed on Schedule B:

- (a) vesting in the portion of the Employer Contribution Account attributable to Employer Contributions made before January 1, 2011 shall be determined in accordance with the Appendix for that Employer that was in effect as of December 31, 2010;
- (b) except as provided on Schedule B, vesting in the portion of the Matching Contribution Account attributable to Matching Contributions made on or after January 1, 2011 shall be determined in accordance with the following vesting schedule subject to the provisos set forth in the first paragraph of this Section 8.2:

Completed Period	
of Service	<u>Vested Percentage</u>
less than 1 year	0%
1	25%
2	50%
3 years or more	100%

(c) except as provided on Schedule B, vesting in the portion of the Supplemental Contribution Account attributable to Supplemental Contributions made on or after January 1, 2011 shall be determined in accordance with the Appendix for that Employer that was in effect as of December 31, 2010.

8.3 Forfeitures

(a) A Participant who incurs a Termination of Employment shall forfeit the nonvested portion of his or her Employer Contribution Account upon the earlier of the date the Participant receives a distribution of his or her vested Account balance or the date the Participant incurs a five-year Period of Severance. Forfeitures shall be applied to reduce contributions that the Employer is required to pay into the Plan and to pay Plan expenses.

(b) If a Participant incurs a forfeiture under subsection (a) and subsequently resumes employment with the Employer or an Affiliate before incurring a five-year Period of Severance, the forfeited amount shall be restored if the Participant repays to the Trust an amount equal to his or her earlier distribution from those Accounts. Such a repayment must be made before the date that is 30 days after the fifth anniversary of the Participant's re-employment date.

ARTICLE IX WITHDRAWALS PRIOR TO TERMINATION OF SERVICE; LOANS

9.1 WITHDRAWALS

A Participant may make a withdrawal from his or her Accounts by providing notice to the Recordkeeper, in accordance with the provisions of this Article IX and the procedures established by the Committee and the Recordkeeper.

9.2 WITHDRAWAL OF AFTER-TAX CONTRIBUTIONS

A Participant may elect to withdraw part or all of the amount credited to his or her After-Tax Contribution Account at any time.

9.3 WITHDRAWAL OF ROLLOVER CONTRIBUTION ACCOUNT

A Participant who has withdrawn the maximum amount permitted under Section 9.2 may elect to withdraw part or all of the amount credited to his or her Rollover Contribution Account at any time.

9.4 WITHDRAWAL OF VESTED MATCHING CONTRIBUTION ACCOUNT

A Participant who has attained age 55, and has withdrawn the maximum amount permitted under Sections 9.2 and 9.3 may withdraw all or a part of the amount credited to his or her vested Matching Contribution Account and Supplemental Contribution Account.

9.5 WITHDRAWAL OF PRE-TAX CONTRIBUTIONS

- (a) A Participant who has attained age 59½ and has withdrawn the maximum amount permitted under Sections 9.2, 9.3, and 9.4 may withdraw part or all of the amount credited to his or her Pre-Tax Contribution Account.
- (b) A Participant who has not attained age 59½ may withdraw part or all of the amount credited to his or her Pre-Tax Contribution Account only as provided in Section 9.6.

9.6 Hardship Withdrawals

(a) A Participant who has not attained age 59½ may take a hardship withdrawal of part or all of the amount credited to his or her Pre-Tax Contribution Account and Catch-Up Contribution Account (but not the earnings on Pre-Tax Contributions or Catch-Up Contributions made after December 31, 1988), but only to the extent required to relieve such financial hardship. No such withdrawal shall be permitted unless the Participant has previously or concurrently withdrawn all amounts available under Sections 9.2 through 9.4 and taken any loans available under Section 9.10. For purposes of this Section, a withdrawal is on account of "hardship" only if the distribution is made on account of an immediate and heavy financial need of the Participant, and such distribution is necessary to satisfy such financial need (including the payment of federal, state and local income taxes and penalties resulting from the hardship withdrawal). A withdrawal will be deemed to be made on account of an immediate and heavy financial need if the withdrawal is on account of:

- (1) unreimbursed expenses for medical care, as defined in Section 213(d) of the Code, incurred by the Participant, his or her spouse, children or dependents;
- (2) purchase (excluding mortgage payments) of the principal residence of the Participant;
- (3) payment of tuition, related educational fees and room and board expenses for the next 12 months of post secondary education for the Participant, his or her spouse, children or dependents;
- (4) the need to prevent the eviction of the Participant from his or her principal residence or foreclosure of the mortgage on the Participant's principal residence;
- (5) funeral or burial expenses for the Participant's deceased parent, spouse, children or dependents;
- (6) expenses for the repair of damage to the Participant's principal residence that would qualify for the casualty deduction under Section 165 of the Code (determined without regard to whether such loss exceeds 10 percent of adjusted gross income); or
- (7) such other events permitted under Section 401(k) of the Code.
- (b) A withdrawal will not be treated as necessary to satisfy an immediate and heavy financial need of a Participant to the extent that the amount of the withdrawal is in excess of the amount required to relieve the financial need or to the extent such need may be satisfied from other resources reasonably available to the Participant, as shall be determined by the Committee in a uniform and non-discriminatory manner on the basis of all the relevant facts and circumstances. A distribution will be deemed necessary to satisfy an immediate and heavy financial need of the Participant if the Committee relies on the Participant's written representation that the need cannot be relieved:
 - (1) through reimbursement or compensation by insurance or otherwise;
 - (2) by reasonable liquidation of the Participant's assets (or those of his or her spouse or minor children) to the extent such liquidation does not create a financial hardship;

- (3) by the Participant's cessation of elective and voluntary contributions under the Plan;
- (4) by the Participant making other withdrawals (including electing distribution of dividends from the ESOP under Section 15.2) or nontaxable loans from all plans in which he or she participates; or
- (5) by borrowing from commercial sources on reasonable commercial terms.
- (c) A Participant may not make any Pre-Tax Contributions, Catch-Up Contributions or After-Tax Contributions for the six-month period following receipt of the hardship distribution.

9.7 WITHDRAWAL OF CATCH-UP CONTRIBUTIONS

A Participant who has attained age 59½ and has withdrawn the maximum amount permitted under Sections 9.2, 9.3, 9.4 and 9.5 may withdraw part or all of the amount credited to his or her Catch-Up Contribution Account.

9.8 WITHDRAWAL PRO-RATA FROM INVESTMENT FUNDS

The amount withdrawn by a Participant under this Article shall be charged on a pro rata basis against the Investment Funds in which the Accounts from which the withdrawal is made are invested.

9.9 TIMING OF WITHDRAWAL PAYMENTS

- (a) In the case of a withdrawal under Sections 9.2 through 9.5 or Section 9.7, the amount withdrawn will be paid to the Participant in a lump sum in cash as soon as practicable following the date of the withdrawal request.
- (b) In the case of a hardship withdrawal under Section 9.6 the amount withdrawn will be paid to the Participant in a lump sum in cash as soon as practicable following approval of the withdrawal.
- (c) No withdrawal of any type is available to Beneficiaries, Alternate Payees (as defined in Plan Section 14.4), or Former Participants.

9.10 <u>Loans</u>

A Participant may take a loan from his or her Accounts by making an application with the Recordkeeper in accordance with procedures established by the Committee and the Recordkeeper.

(a) The maximum amount of any such loan shall be the lesser of (1) \$50,000 reduced by the highest outstanding balance of any loan from the Plan during the one-year period ending on the day before the date on which such loan is made, or (2) 50% of the value of the Participant's vested Account balance under the Plan.

- (b) The minimum amount of any such loan shall be \$1,000. A Participant may have one loan outstanding at any time, provided, however, a Participant may more than one loan outstanding if the Participant is covered by a collective bargaining agreement that so provides or the Participant was a participated in a plan of a predecessor employer which plan transferred more than one loan on behalf of the Participant to this Plan.
- (c) A loan will be made from the Participant's Accounts in the order determined by administrative procedures and from the Investment Funds in which such Accounts are invested on a pro-rata basis. Immediately upon the loan being made, the Participant's Account balance shall be reduced to reflect the outstanding loan balance. All repayments of principal and interest on the Participant's note shall be invested in the Investment Funds in accordance with the Participant's investment election which is in effect at the time of the repayment. If no election is in effect, or made, the repayments of principal and interest shall be invested in a fund that has as its objective the preservation of capital, as determined by the Committee.
- (d) The note for any loan under subsection (a) shall bear interest at a reasonable rate as shall be determined by the Committee; provided, however, that such rate shall not exceed the maximum rate permitted by law. Principal and interest under any such loan shall be repaid by any Participant who is an active Employee through payroll deductions; provided, however, that the Participant may prepay the entire unpaid principal and accrued interest on any loan at any time. The term of such note shall not be for a period longer than five years; provided, however, that, if the proceeds of such loan are used to acquire the Participant's principal residence, the term of such note shall not be for a period longer than 30 years. Loan repayments while a Participant is on "qualified military service," as defined in Code Section 414(u)(5), will be suspended in accordance with Code Section 414(u).
- (e) Any loan to a Participant shall be secured by such Participant's vested interest in his or her Accounts hereunder. As a condition of any such loan, the Participant shall consent to such security interest.
- (f) A Participant who terminates employment may continue to repay any outstanding loan in accordance with procedures established by the Recordkeeper.
- (g) Each Participant to whom a loan is made shall receive a statement of any administrative charges involved in such loan. This statement shall include the dollar amount and annual interest rate of the finance charge. Such administrative charges may be changed within the sole discretion of the Committee, without formal Plan amendment. Such charges will be deducted from the borrower's Account balance.

ARTICLE X - DISTRIBUTIONS

10.1 PAYMENT UPON TERMINATION OF EMPLOYMENT

- (a) A Participant may elect to receive his or her vested account balance in either (1) a lump sum, or (2) monthly, quarterly or annual installments over a period that is at least five years and not more than 20 years, or (3) a combination of the above.
- (b) If a Participant dies before benefit payments have begun, the Participant's vested account balance shall be payable to the Participant's Beneficiary in a lump sum. If the Participant dies after installment payments have begun, the Participant's Beneficiary shall continue to receive the installment payments over the remaining period of time elected by the Participant, provided, however, that the Beneficiary may elect to receive the remaining vested Account balance in a lump sum.

10.2 Cash-Out

Notwithstanding any other provision of this Plan to the contrary, if the Participant's vested Account balance does not exceed \$1,000, the vested Account balance shall be paid to the Participant in a lump sum as soon as practicable following the Participant's Termination of Employment, or to the Participant's Beneficiary following the Participant's death.

10.3 Application for Benefits

Except as provided in Section 10.2, no benefits shall be paid to a Participant until an application therefor shall be made to the Committee. Each application for benefits shall be made with the Recordkeeper in accordance with procedures established by the Committee and the Recordkeeper.

10.4 GENERAL RULES

Notwithstanding any other provision of the Plan to the contrary:

- (a) Subject to making an application in accordance with Section 10.3, the payment of benefits to a Participant or Beneficiary (in the event of the Participant's death) shall be made not later than the 60th day after the later of (1) the close of the Plan Year in which the Participant's Termination of Employment occurs, (2) the close of the Plan Year in which the Participant's 65th birthday occurs, or (3) the 10th anniversary of the year in which the Participant began participation in the Plan.
- (b) Payment of benefits to a Participant shall commence no later than April 1 following (1) the year in which the Participant attains age 70½ or, (2) in the case of a Participant who is not a 5% owner of the Company (or Affiliate), the year in which the Participant retires, in the minimum amount required under Section 401(a)(9) of the Code.

- (c) If the Participant dies before payment of his or her benefits commence, the Participant's entire interest in his or her Accounts shall be paid within five years of the Participant's death to the Participant's Beneficiary.
- (d) Section 10.4(b) and (c) and those provisions of Appendix A that require the Plan to make "required minimum distributions" to participants who have attained age 70-1/2 shall not apply for calendar year 2009. However, the "required beginning date" with respect to any individual shall be determined without regard to this sub-section (d) for purposes of applying Section 10.4(b) and (c) and Appendix A for calendar years after 2009. The five-year period described in Section 10.4(c) and Appendix A, Section 4(b), shall be determined without regard to 2009. This sub-section (d) shall not apply to any required minimum distribution for 2008 that is permitted to be made in 2009 by reason of an individual's required beginning date being April 1, 2009, but it shall apply to any required minimum distribution for 2009 that is permitted to be made in 2010 by reason of an individual's required beginning date being April 1, 2010.

10.5 Consent for Early Distributions

Payment of benefits to a Participant whose vested Account balance exceeds \$1,000 shall not be made prior to the Participant's Normal Retirement Date without the written consent of the Participant.

10.6 <u>Direct Rollover</u>

Notwithstanding any provision of the Plan to the contrary, a Distributee may elect, at the time and in the manner prescribed by the Committee, to have any portion of an Eligible Rollover Distribution paid directly to an Eligible Retirement Plan specified by the Distributee. As used in this Section, the following terms shall have the meanings set forth below:

- (a) "<u>Distributee</u>" means a person who is (1) an Employee or former Employee, (2) the surviving spouse of an Employee or former Employee, or (3) the spouse or former spouse of an Employee or former Employee who is the "alternate payee" under a "qualified domestic relations order", as those terms are defined in Section 414(p) of the Code. A "Distributee" also includes the Employee's non-spouse designated Beneficiary under Section 1.5 of the Plan. In the case of a non-spouse Beneficiary, the direct rollover may be made only to an individual retirement account or annuity described in Section 408(a) or Section 408(b) of the Code ("IRA") that is established on behalf of the designated Beneficiary and that will be treated as an inherited IRA pursuant to the provisions of Section 402(c)(11) of the Code.
- (b) "Eligible Retirement Plan" means an individual retirement account described in Section 408(a) of the Code, an individual retirement annuity described in Section 408(b) of the Code, an annuity plan described in Section 403(a) of the Code, or a qualified trust described in Section 401(a) of the Code, that accepts the Distributee's Eligible Rollover Distribution. An eligible retirement plan shall also mean an annuity contract described in section 403(b) of the Code and an eligible plan under section 457(b) of the Code which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state and which agrees to separately account for amounts transferred into such plan from this Plan. An "Eligible Retirement Plan" shall also include a Roth IRA described in Section 408A of the Code.

(c) "Eligible Rollover Distribution" means any distribution (or withdrawal) of all or any portion of the balance to the credit of the Distributee, except that an Eligible Rollover Distribution does not include any distribution that is one of a series of substantially equal periodic payments made (not less frequently than annually) for the life (or life expectancy) of the Distributee or the joint lives (or joint life expectancies) of the Distributee and the Distributee's designated beneficiary, or for a specified period of ten years or more, any distribution to the extent such distribution is required under Section 401(a)(9) of the Code, the portion of any distribution that is not includible in gross income (determined without regard to the exclusion for net unrealized appreciation with respect to employer securities), or a hardship withdrawal under Section 9.6, or any other distribution that is reasonably expected to total less than \$200 during the year. A portion of a distribution shall not fail to be an eligible rollover distribution merely because the portion consists of After-Tax Contributions provided, however, such portion may be transferred only to an individual retirement account or annuity described in Section 408(a) or (b) of the Code that agrees to separately account for amounts so transferred, including separately accounting for the portion of such distribution that is includible in gross income and the portion of such distribution that is not includible in gross income.

10.7 DISTRIBUTIONS IN CASH OR STOCK

Distributions to a Participant or Beneficiary in the form of a lump sum or installments shall be in cash, provided that, to the extent that distribution is from the Participant's or Beneficiary's ESOP Account invested in the L3 Stock Fund, the portion invested in such Investment Fund shall be distributed in the form of cash or full shares of L3 Stock, at the election of the Participant, with fractional shares paid in cash.

10.8 Qualified Joint and Survivor Annuity

(a) If an applicable Appendix provides for, and the Participant elects, an annuity form of payment, the Participant's vested Account balance shall be used to purchase a Qualified Joint and Survivor Annuity for the Participant. With respect to a Participant who is married on the Annuity Starting Date, a Qualified Joint and Survivor Annuity is an annuity for the life of the Participant and, after the Participant's death, an annuity for the life of the Participant's spouse, in a monthly amount that is 50 percent of the monthly amount paid to the Participant before his or her death. With respect to a Participant who is not married on the Annuity Starting Date, a Qualified Joint and Survivor Annuity is an annuity for the life of the Participant.

- (b) A Participant may waive the Qualified Joint and Survivor Annuity form of payment and elect an alternative form of payment provided under the Plan, including without limitation an annuity for the life of the Participant and, after the Participant's death, an annuity for the life of the Participant's spouse in a monthly amount that is 75 percent of the monthly amount paid to the Participant before his or her death. Any election to waive the Qualified Joint and Survivor Annuity must be made by the Participant in writing during the election period and be consented to by the Participant's spouse. Such spouse's consent must acknowledge the effect of such election and be witnessed by a notary public. Such consent shall not be required if it is established to the satisfaction of the Committee that the required consent cannot be obtained because there is no spouse, the spouse cannot be located, or other circumstances that may be prescribed by Treasury Regulations.
- (c) The election made by the Participant and consented to by the Participant's spouse may be revoked by the Participant in writing without the consent of the spouse at any time during the election period. Any new election must comply with the requirements of subsection (b). A former spouse's waiver shall not be binding on a new spouse.
- (d) The election period to waive the Qualified Joint and Survivor Annuity shall be the 90-day period ending on the Annuity Starting Date. The Annuity Starting Date means the first day of the first period for which an amount is received as an annuity.
- (e) Within a reasonable period of time before the Annuity Starting Date (and consistent with Treasury Regulations), the Participant shall be provided with a written explanation of the terms and conditions of the Qualified Joint and Survivor Annuity, the Participant's right to make an election to waive the Qualified Joint and Survivor Annuity, the right of the Participant's spouse to consent to any election to waive the Qualified Joint and Survivor Annuity, and the right of the Participant to revoke such election and the effect of such revocation.

10.9 QUALIFIED PRERETIREMENT SURVIVOR ANNUITY

- (a) If an applicable Appendix provides for, and the Participant elects, an annuity form of payment and the Participant dies before the Annuity Starting Date, the Participant's vested Account balance shall be used to purchase a Qualified Preretirement Survivor Annuity for the Participant's spouse. A Qualified Preretirement Survivor Annuity is an annuity for the life of the Participant's spouse.
- (b) A Participant may waive the Qualified Preretirement Survivor Annuity form of payment. Any election to waive the Qualified Preretirement Survivor Annuity must be consented to by the Participant's spouse in the same manner provided for in Section 10.8(b) and (c). A Participant may revoke a waiver at any time before the payment of benefits commences without the consent of the spouse, provided that a new waiver shall require a new spousal consent.

ARTICLE XI - SPECIAL TOP-HEAVY PROVISIONS

11.1 TOP-HEAVY RULES

In the event the Plan is or becomes Top-Heavy (as defined in Section 11.2 hereof) in any Plan Year, the provisions of this Article shall apply and shall supersede any conflicting provisions in the Plan for such Plan Year.

11.2 <u>Definitions</u>

As used in this Article, the following terms shall have the meanings set forth below:

- (a) "Determination Date" means with respect to any Plan Year the last day of the preceding Plan Year, and for the first Plan Year, the first day of such Plan Year.
- (b) "Key Employee" means an Employee or former Employee (including a deceased employee) of the Company or an Affiliate who, at any time during the Plan Year that includes the Determination Date was an officer having greater than \$130,000 (as adjusted under Section 416(i)(1) of the Code for Plan Years beginning after December 31, 2002), a 5-percent owner of the Company or an Affiliate, or a 1-percent owner of the Company or an Affiliate. For this purpose, annual compensation means compensation within the meaning of Section 415(c)(3) of the Code. The determination of who is a Key Employee will be made in accordance with Section 416(i)(1) of the Code and the applicable regulations and other guidance of general applicability issued thereunder.
- (c) "Non-Key Employee" means any employee who is not a Key Employee and includes an employee who is a former Key Employee.
- (d) This Plan shall be "Top-Heavy" for any Plan Year if the provisions of any of the following apply:
 - (1) the Top-Heavy Ratio for the Plan exceeds 60% and the Plan is not part of any Required Aggregation Group of Plans or Permissive Aggregation Group of Plans;
 - (2) the Plan is a part of a Required Aggregation Group of Plans (but is not part of a Permissive Aggregation Group of Plans) and the Top-Heavy Ratio for the Required Aggregation Group of Plans exceeds 60%; or
 - (3) the Plan is a part of a Required Aggregation Group of Plans and part of a Permissive Aggregation Group of Plans and the Top-Heavy Ratio for the Permissive Aggregation Group of Plans exceeds 60%.
- (e) "Top-Heavy Ratio" means a fraction: (1) the numerator of which is the sum of the amount credited to accounts under the Plan and any other defined contribution plan maintained by the Company or an Affiliate which is required or permitted to be taken into account for all Key Employees and the Present Value of accrued benefits under any defined benefit plan maintained by the Company or Affiliate which is required or permitted to be taken into account for all Key Employees, and (2) the denominator of which is the sum of the amount credited to the accounts under such defined contribution plans for all Participants and the Present Value of accrued benefits under such defined benefit plans for all Participants. In determining the Top-Heavy Ratio, a Participant's accrued benefit in a defined benefit plan must be determined using the method uniformly used for accrual purposes for all plans of the Company and Affiliate. If there is no such uniform method, the accrued benefit is to be determined as if it accrued not more rapidly than under the slowest rate of accrual permitted under Code Section 411(b)(1)(C).

For purposes of this definition: (A) the amount credited to accounts and the Present Value of accrued benefits shall be determined as of the last day of the most recent Plan Year that falls within or ends with the 12-month period ending on the Determination Date; (B) the amount credited to the accounts and accrued benefits of a Participant who is a Non-Key Employee but who was a Key Employee in a prior year will be disregarded; (C) the amount credited to the accounts and accrued benefits of any individual who has not performed services for the Employer or an Affiliate for the one-year period ending on the Determination Date shall not be taken into account; and (D) the present value of accrued benefits and the account balances of an employee as of the Determination Date shall be increased by the distributions made with respect to the employee under the Plan and any plan aggregated with the Plan under section 416(g)(2) of the Code during the 1-year period ending on the Determination Date. The preceding sentence shall also apply to distributions under a terminated plan which, had it not been terminated, would have been aggregated with the Plan under section 416(g)(2)(A)(i) of the Code. In the case of a distribution made for a reason other than severance from employment, death, or disability, this provision shall be applied by substituting "5-year period" for "1-year period."

- (f) "Required Aggregation Group of Plans" means (1) each qualified plan of the Company or an Affiliate (including a terminated plan) in which at least one Key Employee participates, and (2) any other qualified plan of the Company or an Affiliate which enables a plan described in (1) to meet the requirements of Section 401(a)(4) or 410 of the Code.
- (g) "<u>Permissive Aggregation Group of Plans</u>" means the Required Aggregation Group of Plans plus any other plan or plans of the Company or an Affiliate which, when considered as a group with the Required Aggregation Group of Plans, would continue to satisfy the requirements of Sections 401(a)(4) and 410 of the Code.
- (h) "Present Value" of accrued benefits under any defined benefit plan maintained by the Company or an Affiliate shall mean an actuarial equivalent lump sum amount based on the Pension Benefit Guaranty Corporation factors and assumptions.

11.3 <u>Minimum Contribution</u>

- Except as otherwise provided in subsection (b), for any Plan Year in which the Plan is Top-Heavy, the Employer contributions (and forfeitures) allocated on behalf of any Participant who is a Non-Key Employee (exclusive of any Pre-Tax Contributions on his behalf) shall not be less than 3% of such Participant's Section 415 earnings (as defined in Section 5.5(d) hereof) for such Plan Year. However, should the sum of the Employer's contributions, including Pre-Tax Contributions, and forfeitures allocated to the Account of each Key Employee for such Top-Heavy Plan Year be less than 3% of each Key Employee's Compensation, the sum of the Employer's contributions and forfeitures allocated to the Account of each Non-Key Employee shall be equal to the largest percentage allocated to the Account of each Key Employee. The percentage allocated to the Account of any Key Employee shall be equal to the ratio of the sum of the Employer's contribution and forfeitures allocated on behalf of such Key Employee divided by the Compensation for such Key Employees. The minimum allocation provided for in this Section shall be determined without regard to any contribution to or benefit payable under the Social Security law and shall apply even though under other Plan provisions the Participant would not otherwise be entitled to receive an allocation or would have received a lesser allocation for the applicable Plan Year for any reason. Matching Contributions shall be taken into account for purposes of satisfying the minimum contribution requirements of Section 416(c)(2) and this Section. Matching Contributions that are used to satisfy the minimum contributions requirements shall be treated as matching contributions for purposes of the actual contribution percentage test and other requirements of Section 401(m) of the Code.
- (b) The minimum allocation provided in subsection (a) shall not apply to any Participant who was not an Employee on the last day of the applicable Plan Year or to any Participant to the extent such Participant is covered under any other plan of the Company or an Affiliate which provides for the minimum allocation of Employer contributions and/or accrual of retirement benefits.

11.4 TOP-HEAVY VESTING SCHEDULE

(a) Effective as of the first day of the first Plan Year in which this Plan is Top-Heavy (the "Top-Heavy Effective Date"), the nonforfeitable interest of each Participant in the portion of his or her Employer Contribution Account shall be determined as follows, provided, however, that if the Appendix provides a faster vesting schedule, such vesting schedule shall continue to apply:

Completed Years of Vesting Service	Nonforfeitable Interest
2	20%
3	40%
4	60%
5	80%
6 years or more	100%

(b) Such vesting schedule shall remain in effect for all Plan Years commencing on and after the Top-Heavy Effective Date even though the Plan may not be Top-Heavy for any such Plan Year. Notwithstanding the foregoing provisions of this Section, this Section shall not apply to the benefit of any Participant whose Termination of Employment occurred prior to the Top-Heavy Effective Date.

ARTICLE XII - FUNDING OF THE SAVINGS PLAN; TRUST FUND

12.1 Trust Agreement

The Company has entered into the Trust Agreement with the Trustee to provide for the establishment of a Trust Fund to fund the benefits of the Plan.

12.2 Income on Funds

- (a) The Trust Fund shall consist of the Investment Funds.
- (b) Except as provided in Section 15.2, all dividends and other income, as well as any cash received from the sale or exchange of securities, produced by each Investment Fund shall be reinvested in each such Investment Fund.

12.3 EXCLUSIVE BENEFIT OF TRUST FUND

The principal and income of the Trust Fund shall be used for the exclusive purposes of providing benefits to Participants and their Beneficiaries and defraying reasonable expenses of administering the Plan.

12.4 MISTAKE OF FACT

If a contribution is made to the Plan by the Employer by reason of a mistake of fact, the Employer shall be entitled to receive a return of such contribution, without any gains and net of any losses attributable thereto within one year after making such contribution.

12.5 Contributions Disallowed by Code

All contributions by the Employer to the Plan are conditioned upon the deductibility of such contributions under Section 404 of the Code for the taxable year for which made, and the Employer shall be entitled to receive a return of any contribution, without any gains and net of any losses attributable thereto, to the extent its deduction is disallowed, within one year after such disallowance.

ARTICLE XIII - AMENDMENT AND TERMINATION

13.1 PLAN AMENDMENTS

The Company, by action of the Board of Directors, may at any time modify or amend the Plan, in whole or in part, provided, however, that no such amendment shall make it possible for any of the assets of the Trust Fund to be used for or diverted to purposes other than for the exclusive benefit of Participants and their Beneficiaries, or cause a cut-back in Participants' benefits under the Plan within the meaning of Section 411(d)(6) of the Code. Any such amendment shall be by an instrument in writing approved by the Board of Directors and executed by an officer who is authorized by the Company to sign amendments to the Plan. To the extent permitted by resolution of the Board of Directors, any delegate of the Board may amend this Plan in whole or in part at any time or from time to time. Any such amendment shall be by an instrument in writing.

13.2 PLAN TERMINATION; DISCONTINUANCE OF CONTRIBUTIONS

Although the Company intends to continue the Plan indefinitely, it may, by action of the Board of Directors, discontinue contributions under the Plan or terminate the Plan in part or in its entirety. Any action to terminate the Plan shall be by an instrument in writing executed by the Board of Directors.

13.3 <u>Vesting on Plan Termination</u>

As of the effective date of any termination or partial termination of, or complete discontinuance of contributions to, the Plan, all affected Participants shall become fully vested in their Accounts.

13.4 <u>Distributions on Plan Termination</u>

Upon termination of the Plan, all assets remaining in the Trust Fund, after payment of any expenses properly chargeable against the Trust Fund, shall be distributed to the applicable Participants or their Beneficiaries in accordance with the value of such Participants' Accounts and in accordance with the provisions of the Plan; provided, however, that any amount allocated to a suspense account maintained pursuant to Section 415 of the Code shall be returned to the Company.

ARTICLE XIV - GENERAL PROVISIONS

14.1 No Contract of Employment

Nothing contained in the Plan shall be construed as a contract of employment between the Employer or the Company and any Employee, and the Plan shall not afford an Employee a right to continued employment with the Employer or the Company.

14.2 PAYMENTS SOLELY FROM TRUST FUND

All benefits payable under the Plan shall be paid or provided for solely from the Trust Fund, and neither the Company nor any Employer assumes any liability or responsibility for any Plan payment.

14.3 Incompetency

If the Committee determines that any person to whom a payment is due under the Plan is a minor or is incompetent by reason of physical or mental disability, the Committee shall have the power to cause the payments becoming due to such person to be made to another person or entity, for the benefit of the minor or incompetent, without responsibility of the Company, the Employer, the Committee or the Trustee to see to the application of such payment. Payments made pursuant to such power shall operate as a complete discharge of the Company, the Employer, the Committee, the Trustee and the Trust Fund.

14.4 ALIENATION AND QDROS

- (a) Except as provided below, the interest herein, whether vested or not, of any Participant, Former Participant or Beneficiary, shall not be subject to alienation, assignment, pledge, encumbrance, attachment, garnishment, including, but not limited to, execution, sequestration, or other legal or equitable process, or transferability by operation of law in the event of bankruptcy, insolvency or otherwise.
- (b) The provisions of this Section shall not prevent the creation, assignment or recognition of any individual's right to a benefit payable with respect to a Participant pursuant to a Qualified Domestic Relations Order ("QDRO"). A QDRO shall mean any judgment, decree or order which meets the basic requirements of Code Section 414(p) and meets the QDRO requirements set out in the Plan procedures, concerning domestic relations orders, as determined by the final, discretionary authority of the Committee.
- (c) The Committee shall establish reasonable procedures to determine whether a domestic relations order is a QDRO and to administer distributions under a QDRO. If any domestic relations order is received by the Plan, the Committee shall promptly notify the Participant and any Alternate Payee that the order has been received and of the Plan's procedures for determining whether the order is a QDRO and notify the Participant and each Alternate Payee (or their representatives) of the Committee's determination. "Alternate Payee" shall mean any spouse, former spouse, child or other dependent of a Participant recognized by a proper domestic relations order as having a right to receive all, or a portion of, a Participant's benefits under the Plan, as prescribed under Code Section 414(p).

14.5 Notice to the Committee

If any provision in the Plan describes an Employee or Beneficiary's election, application, or notice to the Committee, then any such action shall only be effective if it is properly made under Plan procedures. Any election, application or notice required to be made shall be deemed to have been made or given on the date received by the Committee or its Recordkeeper.

14.6 Mergers and Transfers

The Board of Directors shall have the power to fully or partially merge the Plan with any other tax-qualified plan or transfer assets to, or accept assets from, any other tax-qualified plan. In the event of any merger or consolidation of the Plan with, or a transfer of the assets and liabilities of the Plan to, any other plan, each Participant shall receive a benefit under such other plan (if such other plan were terminated immediately after such merger, consolidation or transfer) which is equal to or greater than the benefit the Participant would have been entitled to receive under the Plan (if the Plan had been terminated immediately prior to such merger, consolidation or transfer).

14.7 FIDUCIARIES

Any person or group of persons may serve in more than one fiduciary capacity with respect to the Plan hereunder.

14.8 PLANS SHALL COMPLY WITH LAW; CHOICE OF LAW

It is intended that the Plan hereunder conform to and meet the applicable requirements of ERISA and the Code. Except to the extent preempted by ERISA, the validity of the Plan hereunder or of any of the provisions thereof shall be determined under, and they shall be construed and administered according to, the laws of the State of New York, (including its statute of limitations and all substantive and procedural law, and without regard to its conflict of laws provisions). The illegality of any particular provision of the Plan shall not affect the other provisions thereof, but the Plan shall be construed in all respects as if such invalid provision were omitted.

14.9 ERISA 404(c)

The Plan is intended to comply with ERISA Section 404(c). Participants are solely responsible for their own investment choices.

14.10 Gender

The masculine pronoun shall be deemed to include the feminine, and the singular number shall be deemed to include the plural unless a different meaning is plainly required by the context.

14.11 DEEMED DISTRIBUTIONS OF UNVESTED AMOUNTS

In the event of a Participant's Termination of Employment before he or she has any vested interest in his or her Employer Contribution Account, if any, the Participant shall be deemed to have received a distribution of his or her balance as of the Termination of Employment date, in the amount of the unvested portion of his or her Employer Contribution Account. The amount of this deemed distribution shall be zero. Following this deemed distribution, the Participant shall have no remaining benefit under the Plan attributable to his or her Employer Contribution Account.

14.12 HEADINGS

Section headings are provided only for the convenience of the reader. Section headings shall not be considered in interpreting this document.

14.13 MISSING PAYEES

A Participant (or, if deceased, his or her Beneficiary if entitled to Benefits under the Plan) is obligated to keep the Plan Administrator informed as to his or her current address at all times. In the event that a Participant or Beneficiary or other recipient of Benefits cannot be located with reasonable efforts by the end of the second calendar year following the date when Benefits are first payable under the Plan, an amount equal to the Benefit payable may be forfeited. If the Participant or Beneficiary or other recipient of Benefits subsequently makes a claim for these forfeited Benefits, at any time, then the amount forfeited will be reinstated, without interest, and paid as soon as practicable.

14.14 Changes in Vesting Schedule

If the vesting schedule is amended in any way that directly or indirectly affects the computation of the Participant's nonforfeitable percentage such that contributions made subsequent to the amendment will vest more slowly as a result of the amendment, or if the Plan is deemed to be amended by an automatic change to or from a Top-Heavy vesting schedule, each Participant with at least three years of Service with the Employer may elect, within a reasonable period after the adoption of the amendment or change, to have the nonforfeitable percentage computed under the applicable Appendix without regard to such amendment or change. The period during which the election may be made shall commence with the date the amendment is adopted or deemed to be made and shall end on the latest of:

- (a) 60 days after the amendment is adopted;
- (b) 60 days after the amendment becomes effective; or

(c) 60 days after the Participant is issued written notice of the amendment by the Employer or Plan Administrator.

If a vesting schedule is amended, or if the Plan is determined to be Top-Heavy, then the Participants with (1) at least one Hour of Service during the Plan Year for which the change is made and (2) three years of Service with an Employer may elect within a reasonable period, as provided by Code Section 411(a)(10), either the new or former vesting schedule.

14.15 TAX WITHHOLDING

The Committee hereby specifically delegates to the Trustee the responsibility to be liable for income tax withholding, and to withhold the appropriate amount from any payment made from the Trust to any payee under the provisions of applicable law and regulation.

14.16 Common Trust Funds

The Plan adopts and includes the provisions of any group or common trust fund in which the Trust participates, but only as long as such group or common trust fund remains qualified under Section 401(a), and exempt from taxation under Section 501(a), of the Code in accordance with Revenue Ruling 81-100.

ARTICLE XV - ESOP PROVISIONS

15.1 ESOP PORTION OF THE PLAN.

The ESOP is an employee stock ownership plan within the meaning of Code Section 4975(e)(7). The ESOP is maintained as a portion of the Plan as authorized by Treasury Regulations Section 54.4975-11(a)(5). Each Participant's After-Tax Contribution Account, Catch-Up Contribution Account, Matching Contribution Account, Pre-Tax Contribution Account, Rollover Contribution Account, Roth Elective Deferral Account and Supplemental Contribution Account that is invested in whole or in part in the L3 Stock Fund will be divided into an ESOP sub-account and a non-ESOP sub-account will be comprised of that portion of any such Account invested in the L3 Stock Fund and the non-ESOP sub-accounts will together constitute the Participant's or Beneficiary's ESOP Account. All such ESOP Accounts together shall constitute the ESOP. Any reference to the ESOP portion of the Plan shall mean the ESOP Accounts established under the Plan. Unless otherwise specifically stated or unless the context requires otherwise, all other Articles of the Plan and the Trust Agreement shall apply to the Plan as a whole including both the ESOP portion of the Plan and the non-ESOP portion of the Plan.

It is specifically intended that the ESOP acquire and hold L3 Stock that is "qualifying employer securities," as defined in Section 407(d)(5) of ERISA, and "employer securities," as defined in Section 409(l) of the Code, through the L3 Stock Fund.

15.2 <u>Distribution of Dividends</u>

- (a) Cash dividends on L3 Stock attributable to the interests in the L3 Stock Fund allocated to a Participant's or Beneficiary's ESOP Account as of the record date of such dividend shall be paid to the ESOP and held in the L3 Stock Fund unless, in accordance with the Participant's or Beneficiary's Dividend Election, they are distributed in cash to the Participant or Beneficiary no later than 90 days after the last day of the Plan Year in which the dividend is paid.
- (b) The Dividend Election shall be made at such time and in such manner as the Company shall prescribe, provided that Participants and Beneficiaries must be given a reasonable opportunity before a dividend is distributed to make the Dividend Election and must have a reasonable opportunity to change the Dividend Election at least annually. If there is a change in Plan terms governing the manner in which dividends on L3 Stock are paid or distributed to Participants and Beneficiaries, a Participant or Beneficiary must be given a reasonable opportunity to make the Dividend Election under the new Plan terms prior to the date on which the first dividend subject to the new Plan terms is paid or distributed. If a Participant fails to make the Dividend Election, dividends on L3 Stock shall be re-invested in the L3 Stock Fund allocated to the Participant's ESOP Account. The Dividend Election (and the re-investment of dividends as a result of the Participant's failure to make a Dividend Election) shall become irrevocable ten business days before the date that dividends subject to the election are paid to the Plan.

- (c) Notwithstanding any other provision of the Plan to the contrary, dividends on L3 Stock, whether paid to the Participant or Beneficiary or reinvested in the L3 Stock Fund, shall be 100 percent vested at all times, regardless of whether the Participant or Beneficiary is vested in the portion of the L3 Stock Fund with respect to which the dividend is paid. Such dividends shall not be treated as Annual Additions under Section 5.5, Pre-Tax Contributions, After-Tax Contributions, Matching Contributions or Supplemental Contributions under the other provisions of the Plan. Dividends distributed to Participants are not subject to the consent requirements of Section 10.5.
- (d) In order to receive a hardship distribution under Section 9.6, a Participant must have elected under any currently available Dividend Election to receive the dividends in cash.
- (e) Dividends distributed pursuant to this Section are not Eligible Rollover Distributions for purposes of Section 10.6 even if distributed at the same time as other amounts that do constitute Eligible Rollover Distributions.

IN WITNESS WHEREOF, this L3 Technologies Master Savings Plan is hereby amended and restated effective January 1, 2017.

L3 TECHNOLOGIES, INC.

Date: January 2, 2017 By: /s/ Kevin L. Weiss

Title: Corporate Vice President, Human Resources

58

SCHEDULE A MINIMUM REQUIRED DISTRIBUTIONS

Section 1. General Rules.

- (a) <u>Effective Date</u>. Notwithstanding any other provision of the Plan to the contrary, the provisions of this Appendix will apply for purposes of determining required minimum distributions for calendar years beginning with the 2003 calendar year.
- (b) <u>Treasury Regulations Incorporated by Reference</u>. All distributions required under this Appendix will be determined and made in accordance with the Treasury regulations under section 401(a)(9) of the Code.

Section 2. <u>Time and Manner of Distribution.</u>

- (a) Required Beginning Date. The Participant's entire interest will be distributed, or begin to be distributed, to the Participant no later than the Participant's Required Beginning Date.
- (b) <u>Death of Participant Before Distributions Begin.</u> If the Participant dies before distributions begin, the Participant's entire interest will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death. If the Participant's surviving spouse is the Participant's sole Designated Beneficiary and the surviving spouse dies after the Participant but before distributions to the surviving spouse begin, this Section 2(b) will apply as if the surviving spouse were the Participant.
- (c) <u>Forms of Distribution</u>. Unless the Participant's interest is distributed in a single sum on or before the Required Beginning Date, as of the first Distribution Calendar Year, distributions will be made in accordance with Sections 3 and 4 of this Schedule.

Section 3. <u>Required Minimum Distributions During Participant's Lifetime</u>.

- (a) <u>Amount of Required Minimum Distribution For Each Distribution Calendar Year</u>. During the Participant's lifetime, the minimum amount that will be distributed for each Distribution Calendar Year is the lesser of:
 - (1) the quotient obtained by dividing the Participant's Account Balance by the distribution period in the Uniform Lifetime Table set forth in Treas. Reg. §1.401(a)(9)–9, using the Participant's age as of the Participant's birthday in the Distribution Calendar Year; or
 - (2) if the Participant's sole Designated Beneficiary for the Distribution Calendar Year is the Participant's spouse, the quotient obtained by dividing the Participant's Account Balance by the number in the Joint and Last Survivor Table set forth in §1.401(a)(9)–9, using the Participant's and spouse's attained ages as of the Participant's and spouse's birthdays in the Distribution Calendar Year.

A-1

(b) <u>Lifetime Required Minimum Distributions Continue Through Year of Participant's Death</u>. Required minimum distributions will be determined under this Section 3 beginning with the first Distribution Calendar Year and up to and including the Distribution Calendar Year that includes the Participant's date of death.

Section 4. Required Minimum Distributions After Participant's Death.

- (a) <u>Death On or After Date Distributions Begin.</u>
 - (1) <u>Participant Survived by Designated Beneficiary</u>. If the Participant dies on or after the date distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the longer of the remaining Life Expectancy of the Participant or the remaining Life Expectancy of the Participant's Designated Beneficiary, determined as follows:
 - (A) The Participant's remaining Life Expectancy is calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.
 - (B) If the Participant's surviving spouse is the Participant's sole Designated Beneficiary, the remaining Life Expectancy of the surviving spouse is calculated for each Distribution Calendar Year after the year of the Participant's death using the surviving spouse's age as of the spouse's birthday in that year. For Distribution Calendar Years after the year of the surviving spouse's death, the remaining Life Expectancy of the surviving spouse is calculated using the age of the surviving spouse as of the spouse's birthday in the calendar year of the spouse's death, reduced by one for each subsequent calendar year.
 - (C) If the Participant's surviving spouse is not the Participant's sole Designated Beneficiary, the Designated Beneficiary's remaining Life Expectancy is calculated using the age of the beneficiary in the year following the year of the Participant's death, reduced by one for each subsequent year.
 - (2) No Designated Beneficiary. If the Participant dies on or after the date distributions begin and there is no Designated Beneficiary as of September 30 of the year after the year of the Participant's death, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the Participant's remaining Life Expectancy calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.

(b) <u>Death Before Distributions Begin</u>. If the Participant dies before distributions begin, the Participant's entire interest will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death. If the Participant's surviving spouse is the Participant's sole Designated Beneficiary and the surviving spouse dies after the Participant but before distributions to the surviving spouse begin, this Section 4(b) will apply as if the surviving spouse were the Participant.

Section 5. <u>Definitions</u>.

- (a) <u>Designated Beneficiary</u>. The individual who is designated as the beneficiary under the Plan and is the Designated Beneficiary under section 401(a)(9) of the Internal Revenue Code and Treas. Reg. §1.401(a)(9)–1, Q&A-4.
- (b) <u>Distribution Calendar Year</u>. A calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first Distribution Calendar Year is the calendar year immediately preceding the calendar year which contains the Participant's Required Beginning Date. For distributions beginning after the Participant's death, the first Distribution Calendar Year is the calendar year in which distributions are required to begin under Section 2(b). The required minimum distribution for the Participant's first Distribution Calendar Year will be made on or before the Participant's Required Beginning Date. The required minimum distribution for other Distribution Calendar Years, including the required minimum distribution for the Distribution Calendar Year in which the Participant's Required Beginning Date occurs, will be made on or before December 31 of that Distribution Calendar Year.
- (c) <u>Life Expectancy</u>. Life Expectancy as computed by use of the Single Life Table in Treas. Reg. §1.401(a)(9)–9.
- (d) Participant's Account Balance. The Account Balance as of the last Valuation Date in the calendar year immediately preceding the Distribution Calendar Year ("valuation calendar year") increased by the amount of any contributions made and allocated or forfeitures allocated to the Account Balance as of dates in the valuation calendar year after the Valuation Date and decreased by distributions made in the valuation calendar year after the Valuation Date. The Account Balance for the valuation calendar year includes any amounts rolled over or transferred to the Plan either in the valuation calendar year or in the Distribution Calendar Year if distributed or transferred in the valuation calendar year.

(e)	Required Beginning Date. April 1 of the calendar year following the later of (1) the calendar year in which the Participant attains age 70 ½
	or (2) in the case of a Participant who is not a 5% owner of the Company, the year in which the Participant retires.

Simpson Thacher & Bartlett LLP

425 LEXINGTON AVENUE NEW YORK, NY 10017-3954

TELEPHONE: +1-212-455-2000 FACSIMILE: +1-212-455-2502

Direct Dial Number +1-212-455-2000

E-mail Address

February 23, 2017

L3 Technologies, Inc. 600 Third Avenue New York, New York 10016

Ladies and Gentlemen:

We have acted as counsel to L3 Technologies, Inc., a Delaware corporation (the "Company"), in connection with the Registration Statement on Form S-8 (the "Registration Statement") filed by the Company with the Securities and Exchange Commission (the "Commission") under the Securities Act of 1933, as amended (the "Act"), relating to the issuance by the Company of up to 3,000,000 shares of Common Stock (the "Shares"), par value \$.01 per share, which may be issued pursuant to the L3 Technologies Master Savings Plan (the "Plan").

We have examined the Registration Statement, a form of the share certificate and the Plan. We also have examined the originals, or duplicates or certified or conformed copies, of such other records, agreements, documents and other instruments and have made such other investigations as we have deemed relevant and necessary in connection with the opinions hereinafter set forth. As to questions of fact material to this opinion, we have relied upon certificates or comparable documents of public officials and of officers and representatives of the Company.

WASHINGTON,

BEIJING HONG KONG HOUSTON LONDON LOS ANGELES PALO ALTO SÃO PAULO SEOUL TOKYO D.C.

In rendering the opinion set forth below, we have assumed the genuineness of all signatures, the legal capacity of natural persons, the authenticity of all documents submitted to us as originals, the conformity to original documents of all documents submitted to us as duplicates or certified or conformed copies and the authenticity of the originals of such latter documents.

Based upon the foregoing, and subject to the qualifications, assumptions and limitations stated herein, we are of the opinion that, upon issuance and delivery of the Shares in accordance with the Plan, the Shares will be validly issued, fully paid and nonassessable.

We do not express any opinion herein concerning any law other than the Delaware General Corporation Law.

We hereby consent to the filing of this opinion letter as Exhibit 5 to the Registration Statement.

Very truly yours,

/s/ Simpson Thacher & Bartlett LLP

SIMPSON THACHER & BARTLETT LLP

Exhibit 23.1

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We hereby consent to the incorporation by reference in this Registration Statement on Form S-8 of our report dated February 23, 2017 relating to the financial statements, and the effectiveness of internal control over financial reporting, which appears in L3 Technologies, Inc.'s Annual Report on Form 10-K for the year ended December 31, 2016.

/s/ PricewaterhouseCoopers LLP

New York, New York February 23, 2017